# Regional versus general anaesthesia for caesarean section (Review)

Afolabi BB, Lesi FEA



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2012, Issue 10

http://www.thecochranelibrary.com



# TABLE OF CONTENTS

HEADER	1
ABSTRACT	1
PLAIN LANGUAGE SUMMARY	2
BACKGROUND	2
OBJECTIVES	3
METHODS	3
RESULTS	6
DISCUSSION	9
AUTHORS' CONCLUSIONS	10
ACKNOWLEDGEMENTS	10
REFERENCES	10
CHARACTERISTICS OF STUDIES	14
DATA AND ANALYSES	45
Analysis 1.1. Comparison 1 Epidural versus general anaesthesia. Outcome 1 Mean umbilical arterial pH	48
Analysis 1.2 Comparison 1 Epidural versus general anaesthesia. Outcome 2 Mean umbilical venous nH	49
Analysis 1.2. Comparison 1 Epidural versus general anaesthesia, Outcome 2 Angar score of 4 or less at 1 minute	50
Analysis 1.4. Comparison 1 Epidural versus general anaesthesia, Outcome / Apgar score of / or less at 5 minutes	50
Analysis 1.5. Comparison 1 Epidural versus general anaesthesia, Outcome 5 Angar score of 6 or less at 1 minutes.	)0
maysis 1.7. Comparison 1 Epidular versus general anaesticsia, Outcome 7 Apgar score of 0 of less at 1 minute (not	51
Analyzia 1.6 Comparison 1 Enidural variaus concerts and anotherize Outcome 6 Anotherize and 6 for loss at 5 minutes (not	)1
Analysis 1.0. Comparison 1 Epidural versus general anaestnesia, Outcome o Apgal scole of o of less at 5 minutes (not	50
prespecified in protocol).	)2 52
A lai 1.0 Comparison 1 Epidural versus general anaestnesia, Outcome / Niean Apgar score at 1 minute.	)) 5 (
Analysis 1.8. Comparison 1 Epidural versus general anaestnesia, Outcome 8 Mean Apgar score at 5 minutes.	54
Analysis 1.9. Comparison 1 Epidural versus general anaesthesia, Outcome 9 Neonatal neurologic and adaptive capacity	
score at 2-4 hours.	22
Analysis 1.10. Comparison I Epidural versus general anaesthesia, Outcome 10 Amount of blood transfusion received in	
units (not prespecihed in protocol).	55
Analysis 1.11. Comparison 1 Epidural versus general anaesthesia, Outcome 11 Number who received postoperative blood	
transfusion (not prespecified in protocol).	56
Analysis 1.12. Comparison 1 Epidural versus general anaesthesia, Outcome 12 Maternal estimated blood loss in mL.	56
Analysis 1.13. Comparison 1 Epidural versus general anaesthesia, Outcome 13 Difference between pre and postoperative	
haematocrit (%)	57
Analysis 1.14. Comparison 1 Epidural versus general anaesthesia, Outcome 14 Satisfaction score on visual analogue scale.	58
Analysis 1.15. Comparison 1 Epidural versus general anaesthesia, Outcome 15 Number who would prefer the same	
technique again	58
Analysis 1.16. Comparison 1 Epidural versus general anaesthesia, Outcome 16 Neonatal neurologic and adaptive capacity	
score of < 35 at 15 minutes	59
Analysis 1.17. Comparison 1 Epidural versus general anaesthesia, Outcome 17 Neonatal neurologic and adaptive capacity	
score of < 35 at 2 hours	59
Analysis 1.18. Comparison 1 Epidural versus general anaesthesia, Outcome 18 Adverse events	60
Analysis 1.19. Comparison 1 Epidural versus general anaesthesia, Outcome 19 Need for oxygen therapy or mask ventilation	
of the neonate	62
Analysis 1.20. Comparison 1 Epidural versus general anaesthesia, Outcome 20 Maternal blood loss > 500 mL.	62
Analysis 1.21. Comparison 1 Epidural versus general anaesthesia, Outcome 21 Mean Apgar score at 10 minutes.	63
Analysis 1.22. Comparison 1 Epidural versus general anaesthesia, Outcome 22 Time to request postoperative analgesia in	
minutes.	63
Analysis 1.23. Comparison 1 Epidural versus general anaesthesia, Outcome 23 Intraoperative pain score on visual analogue	
scale	64
Analysis 2.1. Comparison 2 Spinal versus general anaesthesia, Outcome 1 Umbilical arterial pH.	65
Analysis 2.2. Comparison 2 Spinal versus general anaesthesia. Outcome 2 Umbilical venous pH.	66
	<del></del>
regional versus general anaestnesia for caesarean section (review)	

1	Analysis 2.3. Comparison 2 Spinal versus general anaesthesia, Outcome 3 Neonatal neurologic and adaptive capacity score	66
1	Analysis 2.4. Comparison 2 Spinal versus general anaesthesia, Outcome 4 Neonatal neurologic and adaptive capacity score	00
	of < 35 at 15 minutes	67
1	Analysis 2.5. Comparison 2 Spinal versus general anaesthesia, Outcome 5 Neonatal neurologic and adaptive capacity score of < 35 at 2 hours.	68
	Analysis 2.6. Comparison 2 Spinal versus general anaesthesia. Outcome 6 Neonatal neurologic and adaptive capacity score	
	of < 35 at 24 hours.	68
	Analysis 2.7. Comparison 2 Spinal versus general anaesthesia. Outcome 7 Mean Apgar score at 1 minute.	69
_	Analysis 2.8. Comparison 2 Spinal versus general anaesthesia, Outcome 8 Mean Apgar score at 5 minutes.	70
_	Analysis 2.9. Comparison 2 Spinal versus general anaesthesia. Outcome 9 Apgar score of 6 or less at 1 minute (not	, -
	prespecified in protocol).	71
1	Analysis 2.10. Comparison 2 Spinal versus general anaesthesia, Outcome 10 Apgar score of 6 or less at 5 minutes (not	
	prespecified in protocol).	72
1	Analysis 2.11. Comparison 2 Spinal versus general anaesthesia, Outcome 11 Maternal estimated blood loss in mL.	73
1	Analysis 2.12. Comparison 2 Spinal versus general anaesthesia, Outcome 12 Difference between pre and postoperative	
	haematocrit.	73
1	Analysis 2.13. Comparison 2 Spinal versus general anaesthesia, Outcome 13 Number who received postoperative blood	
	transfusion (not prespecified in protocol).	74
1	Analysis 2.14. Comparison 2 Spinal versus general anaesthesia, Outcome 14 Number who would prefer the same technique	
	again	74
1	Analysis 2.15. Comparison 2 Spinal versus general anaesthesia, Outcome 15 Satisfaction score on visual analogue scale.	75
1	Analysis 2.16. Comparison 2 Spinal versus general anaesthesia, Outcome 16 Adverse events	76
1	Analysis 2.17. Comparison 2 Spinal versus general anaesthesia, Outcome 17 Maternal blood loss > 500 mL	77
1	Analysis 2.18. Comparison 2 Spinal versus general anaesthesia, Outcome 18 Mean Apgar score at 10 minutes	77
1	Analysis 2.19. Comparison 2 Spinal versus general anaesthesia, Outcome 19 Time to request postoperative analgesia in	
	minutes.	78
1	Analysis 2.20. Comparison 2 Spinal versus general anaesthesia, Outcome 20 Neonatal need for oxygen by mask or intubation.	78
1	Analysis 2.21. Comparison 2 Spinal versus general anaesthesia, Outcome 21 Intraoperative pain score on visual analogue	
	scale	79
1	Analysis 3.1. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 1 Umbilical arterial pH.	79
1	Analysis 3.2. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 2 Apgar score of 6 or less at 1	
	minute (not prespecified in protocol).	80
1	Analysis 3.3. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 3 Apgar score of 6 or less at 5	
	minutes (not prespecified in protocol).	80
1	Analysis 3.4. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 4 Need for oxygen therapy or	
	mask ventilation of neonate.	81
1	Analysis 3.5. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 5 Mean Apgar score at 1	
		82
1	Analysis 3.6. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 6 Mean Apgar score at 5	07
	ITIMALES	82
APPF	NDICES	83
WHA	T'S NFW	84
HIST	ORY	84
CON	TRIBUTIONS OF AUTHORS	85
DEC	LARATIONS OF INTEREST	85
DIFF	ERENCES BETWEEN PROTOCOL AND REVIEW	85
INDE	EX TERMS	86

Regional versus general anaesthesia for caesarean section (Review) Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd. ii

#### [Intervention Review]

# Regional versus general anaesthesia for caesarean section

## Bosede B Afolabi<sup>1</sup>, Foluso EA Lesi<sup>2</sup>

<sup>1</sup>Department of Obstetrics and Gynaecology, University of Lagos, Lagos, Nigeria. <sup>2</sup>Department of Paediatrics and Child Health, College of Medicine of the University of Lagos, Lagos, Nigeria

Contact address: Bosede B Afolabi, Department of Obstetrics and Gynaecology, University of Lagos, College of Medicine, PMB 12003, Idi-Araba, Lagos, Nigeria. bosedeafolabi2003@yahoo.com.

Editorial group: Cochrane Pregnancy and Childbirth Group.

**Publication status and date:** New search for studies and content updated (no change to conclusions), published in Issue 10, 2012. **Review content assessed as up-to-date:** 30 November 2011.

Citation: Afolabi BB, Lesi FEA. Regional versus general anaesthesia for caesarean section. *Cochrane Database of Systematic Reviews* 2012, Issue 10. Art. No.: CD004350. DOI: 10.1002/14651858.CD004350.pub3.

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

# ABSTRACT

#### Background

Regional anaesthesia (RA) and general anaesthesia (GA) are commonly used for caesarean section (CS) and both have advantages and disadvantages. It is important to clarify what type of anaesthesia is more efficacious.

#### Objectives

To compare the effects of RA with those of GA on the outcomes of CS.

#### Search methods

We searched the Cochrane Pregnancy and Childbirth Group's Trials Register (30 November 2011). We updated the search on 20 August 2012 and added the results to the awaiting classification section of the review.

#### Selection criteria

Randomised and quasi-randomised controlled trials evaluating the use of RA and GA in women who had CS for any indication. Cluster-randomised trials and trials using a cross-over design are not included.

#### Data collection and analysis

Two review authors independently assessed trials for inclusion, assessed risk of bias and extracted data. Data were checked for accuracy.

#### Main results

Twenty-two out of 29 included studies (1793 women) contributed data to this review.

The included studies did not report some our primary outcomes: maternal death, incidence of maternal postoperative wound infection, maternal postoperative other infection such as endometritis and urinary tract infection, neonatal death.

Compared to women who had GA, women who had either epidural anaesthesia or spinal anaesthesia were found to have a significantly lower difference between pre and postoperative haematocrit. For epidural, the mean difference (MD) was 1.70% and 95% confidence interval (CI) 0.47 to 2.93 (one trial, 231 women) and for spinal anaesthesia, the MD was 3.10% and 95% CI 1.73 to 4.47 (one trial, 209 women). Compared with GA, women having either an epidural anaesthesia or spinal anaesthesia had a lower estimated maternal blood loss (epidural versus GA: standardised mean difference (SMD) -0.32 mL; 95% CI -0.56 to -0.07; two trials, 256 women; spinal versus GA anaesthesia: SMD -0.59 mL; 95% CI -0.83 to 0.35; two trials, 279 women). There was evidence of a significant difference

in terms of satisfaction with anaesthetic technique - compared with the epidural or spinal group, more women in the GA group stated they would use the same technique again if they needed CS for a subsequent pregnancy (epidural versus GA: risk ratio (RR) 0.80; 95% CI 0.65 to 0.98; one trial, 223 women; spinal versus GA anaesthesia: RR 0.80; 95% CI 0.65 to 0.99; one trial, 221 women).

No significant difference was seen in terms of neonatal Apgar scores of six or less and of four or less at five minutes and the need for neonatal resuscitation with oxygen.

#### Authors' conclusions

There is no evidence from this review to show that RA is superior to GA in terms of major maternal or neonatal outcomes. Further research to evaluate neonatal morbidity and maternal outcomes, such as satisfaction with technique, will be useful.

### PLAIN LANGUAGE SUMMARY

#### Regional versus general anaesthesia for caesarean section

Caesarean section is when a baby is born through an incision in the mother's abdomen and uterine wall. This requires effective anaesthesia which can be regional (epidural or spinal) or a general anaesthetic. With regional epidural anaesthesia, the anaesthetic is infused into the space around the mother's spinal column, whilst with regional spinal anaesthesia, the drug is injected as a single dose into the mother's spinal column. With the two types of regional anaesthesia, the mother is awake for the birth but numbed from the waist down. With general anaesthesia, the mother is unconscious for the birth with the anaesthetic affecting her whole body. As well as women having a view as to whether they might wish to be awake or asleep for the caesarean birth, it is important to know the balance of the benefits and adverse effects of these different types of anaesthesia. This review of trials sought to assess these benefits and harms. Twenty-two out of 29 included studies (1793 women) contributed data to this review. There were some differences that favoured regional anaesthesia, for example, less blood loss. The evidence on the differences in pain relief was difficult to evaluate. There were not enough participants to assess the very rare outcome of mortality for the mother, which may be an important aspect. None of the trials addressed important outcomes for women such as recovery times, effects on breastfeeding, effects on the mother-child relationship and length of time before mother feels well enough to care for her baby. As there is insufficient evidence on benefits and adverse effects, women are most likely to choose anaesthesia for caesarean section, depending on whether they wish to be awake or asleep for the birth.

# BACKGROUND

Caesarean section refers to the procedure where a baby is delivered through an incision on the abdominal wall and uterus of the mother. It is often life-saving and aims to preserve the health of the mother and her baby. Although the operation has become very safe over the years, it is still associated with greater maternal mortality and morbidity (Enkin 2000; Liu 2007). The risk of maternal death with caesarean section is four times that associated with all types of vaginal birth, which is one per 10,000 births (Enkin 2000). It is known that there is a greater risk of neonatal respiratory distress with caesarean section than vaginal delivery, regardless of gestational age (Zanardo 2004; Kolås 2006). This has been described as mild and transient (Danforth 1985) however, and caesarean section is usually considered safe for the fetus. Caesarean section is often described as elective (when it is planned) or emergency.

The type of anaesthesia used and the care with which it is administered is an important determinant of the outcome of caesarean section (Andersen 1987; Enkin 2000). Regional and general anaesthesia are commonly used for caesarean section and both have their advantages and disadvantages (Thorp 2009). General anaesthesia refers to the loss of ability to perceive pain associated with loss of consciousness produced by intravenous or inhalation anaesthetic agents. For caesarean section, this involves the use of thiopentone for induction, tracheal intubation facilitated by suxamethonium, positive-pressure ventilation of the lungs with a nitrous oxide/oxygen mixture plus a volatile agent, and a muscle relaxant (Thorburn 1998). The risks include the aspiration of stomach contents, awareness of the surgical procedure (due to inadequate anaesthesia), failed intubations, and respiratory problems for both mother and baby (Thorp 2009). When supplemented with halogenated volatile agents, general anaesthesia has also been associated with a greater risk of maternal blood loss compared with regional anaesthesia (Andrews 1992). However, it is a more quickly administered procedure and is often preferred in cases

Regional versus general anaesthesia for caesarean section (Review) Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

where speed is important (Thorp 2009). Also recent studies have found no incidence of failed intubation in some large series where general anaesthesia was carried out or supervised by experienced anaesthetists (Djabatey 2009; Ajmal 2011).

Regional anaesthesia refers to the use of local anaesthetic solutions to produce circumscribed areas of loss of sensation. The types of regional anaesthesia used for caesarean section (that is, spinal (subarachnoid) and epidural (extradural) anaesthesia) involve the infiltration of a local anaesthetic agent, usually bupivacaine, into the surroundings of the spinal cord through the lower back of the woman. With spinal anaesthesia, the drug is injected directly into the subarachnoid space while, with epidural, it is injected through a catheter that has been introduced into the extradural space (Thorburn 1998).

Spinal and epidural anaesthesia cause a substantial drop in maternal blood pressure, which may affect both mother and fetus (Visalyaputra 2005; Macarthur 2007), and may be dangerous when the woman has a bleeding complication (Enkin 2000). They are also contraindicated in women with coagulation (clotting) disorders since the insertion of the block may precipitate a bleed. They may cause a severe postdural puncture headache although the incidence of this is now reduced with the use of special needles (Kestin 1991). The advantages of regional anaesthesia include the reduction of the incidence of general anaesthetic complications and that of early bonding between the mother and the newborn, since the mother is awake during the procedure (Enkin 2000). Specifically, spinal and epidural anaesthesia are similar in their safety profiles with a few differences. Spinal anaesthesia has a faster onset of action and requires less of the drug, but causes more hypotensive episodes than epidural anaesthesia (Ng 2004).

Regional anaesthesia is the preferred method for caesarean section in the United Kingdom and the United States of America (USA) (Gibbs 1986; Hibbard 1996). In the USA in particular, regional anaesthesia was used for caesarean section in over 80% of cases as of 1992, regardless of the indication (Hawkins 1997a), and in over 50% of cases as far back as 1981 (Hawkins 1997a). The reasons for this trend have been attributed to the fact that maternal mortality with regional anaesthesia has been reducing steadily over the years while that of general anaesthesia remains the same (Hawkins 1997b), and to the greater familiarity of anaesthesia residents with the procedure (Hawkins 1997a). However, general anaesthesia is still frequently used in some countries, also largely due to greater familiarity with it (Stamer 2005; Furmanik 2010; Ajmal 2011).

The effect on neonates is less clear with some studies showing no difference in neonatal outcome between the two groups (Fox 1979; Zagorzycki 1982) and others maintaining that neonatal outcome is better with regional than with general anaesthesia ( Abboud 1985; Ong 1989). Most of the studies that report no difference are those done on women who had elective operations (Korkmaz 2004) while those done on emergencies tend to report a positive difference in neonatal outcome with regional anaesthesia compared with general (Dyer 2003).

Given the benefits and risks of the different techniques, it is important to clarify what type of anaesthesia is more efficacious in terms of various maternal and neonatal outcomes for the different types of, and indications for, caesarean section.

# OBJECTIVES

To compare the effects of regional anaesthesia with those of general anaesthesia on the outcomes of caesarean section.

# METHODS

#### Criteria for considering studies for this review

#### Types of studies

Randomised and quasi-randomised controlled trials. Cluster-randomised trials and trials using a cross-over design are not included.

#### Types of participants

Mothers having elective or emergency caesarean section for any indication, with the various definitions of elective and emergency taken into consideration.

#### **Types of interventions**

Intervention: regional anaesthesia, whether spinal, epidural or any combination of both.

Control: general anaesthesia using any combination of anaesthetic drugs and muscle relaxants.

#### Types of outcome measures

#### **Primary outcomes**

#### Maternal

#### 1. Maternal death

- 2. Incidence of postoperative wound infection
- 3. Incidence of other postoperative infections such as
- endometritis and urinary tract infection
- 4. Mean difference between pre and postoperative haematocrit or haemoglobin levels
  - 5. Maternal blood loss greater than 500 mL

Regional versus general anaesthesia for caesarean section (Review) Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

6. Mean maternal blood loss

7. Amount of blood transfusion received in units (not prespecified in protocol)

8. Number who received postoperative blood transfusion (not prespecified in protocol)

#### Neonatal

- 1. Neonatal death
- 2. Mean umbilical arterial or venous pH
- 3. Mean neonatal neurologic and adaptive score
- 4. Mean neonatal Apgar scores at one and five minutes

#### Secondary outcomes

#### Maternal

- 1. Incidence of intraoperative pain
- 2. Maternal satisfaction with anaesthetic technique
- 3. Need for postoperative analgesia
- 4. Incidence of postoperative nausea and vomiting

5. Time to request postoperative analgesia in minutes (not prespecified in protocol)

6. Adverse events such as anaphylactic reactions, thromboembolic disease and backache. Headache, epigastric pain, blurred vision, convulsions, pruritus, shivering and bradycardia were also measured despite not being prespecified in the protocol

# Neonatal

- 1. Time to sustained respiration
- 2. Need for oxygen by mask or intubation

3. Apgar score of four or less at one and five minutes (not prespecified in protocol)

4. Apgar score of six or less at one and five minutes (not prespecified in protocol)

5. Apgar score of eight or less at one and five or 10 minutes (not prespecified in protocol

6. Mean neonatal Apgar scores at one and 10 minutes (not prespecified in protocol)

# Search methods for identification of studies

#### **Electronic searches**

We searched the Cochrane Pregnancy and Childbirth Group's Trials Register by contacting the Trials Search Co-ordinator (30 November 2011). We updated this search on 20 August 2012 and added the results to Studies awaiting classification for consideration in the next update.

The Cochrane Pregnancy and Childbirth Group's Trials Register is maintained by the Trials Search Co-ordinator and contains trials identified from:

1. quarterly searches of the Cochrane Central Register of Controlled Trials (CENTRAL);

2. weekly searches of MEDLINE;

3. weekly searches of EMBASE;

4. handsearches of 30 journals and the proceedings of major conferences;

5. weekly current awareness alerts for a further 44 journals plus monthly BioMed Central email alerts.

Details of the search strategies for CENTRAL, MEDLINE and EMBASE, the list of handsearched journals and conference proceedings, and the list of journals reviewed via the current awareness service can be found in the 'Specialized Register' section within the editorial information about the Cochrane Pregnancy and Childbirth Group.

Trials identified through the searching activities described above are each assigned to a review topic (or topics). The Trials Search Co-ordinator searches the register for each review using the topic list rather than keywords.

See Appendix 1 for details of additional searching carried out in the previous version of the review.

We did not apply any language restrictions.

# Data collection and analysis

For the methods used when assessing the trials identified in the previous version of this review, *see* Appendix 2.

For this update we used the following methods when assessing the trials identified by the updated search.

#### Selection of studies

Bosede Afolabi (BA) and Afolabi Lesi (AL) independently assessed for inclusion all the potential studies identified as a result of the search strategy.

### Data extraction and management

We designed a form to extract data. For eligible studies, BA and AL extracted the data using the agreed form. We resolved discrepancies through discussion. We entered data into Review Manager software (RevMan 2011) and checked them for accuracy. In cases when information on the above was unclear, we attempted to contact authors of the original reports to provide further details.

#### Assessment of risk of bias in included studies

BA and AL independently assessed risk of bias for each study using the criteria outlined in the *Cochrane Handbook for Systematic Reviews of Interventions* (Higgins 2011). We resolved all disagreements by discussion.

# (1) Random sequence generation (checking for possible selection bias)

We describe for each included study the method used to generate the allocation sequence in sufficient detail to allow an assessment of whether it should produce comparable groups. We assessed the method as:

we assessed the method as:

• low risk of bias (any truly random process, e.g. random number table; computer random number generator);

• high risk of bias (any non-random process, e.g. odd or even date of birth; hospital or clinic record number);

• unclear risk of bias.

# (2) Allocation concealment (checking for possible selection bias)

We describe for each included study the method used to conceal allocation to interventions prior to assignment and assessed whether intervention allocation could have been foreseen in advance of, or during recruitment, or changed after assignment.

We assessed the methods as:

• low risk of bias (e.g. telephone or central randomisation; consecutively numbered sealed opaque envelopes);

• high risk of bias (open random allocation; unsealed or nonopaque envelopes, alternation; date of birth);

• unclear risk of bias.

# (3) Blinding of participants, personnel (checking for possible performance bias) and outcome assessment (checking for possible detection bias)

We describe for each included study the methods used, if any, to blind study participants and personnel from knowledge of which intervention a participant received. We consider studies to be at low risk of bias if they were blinded, or if we judge that the lack of blinding would be unlikely to affect results. We assessed blinding separately for different outcomes or classes of outcomes. We assessed the methods as:

- low, high or unclear risk of bias for participants;
- low, high or unclear risk of bias for personnel;
- low, high or unclear risk of bias for outcome assessment.

(4) Incomplete outcome data (checking for possible attrition bias due to the amount, nature and handling of incomplete outcome data) We describe for each included study, and for each outcome or class of outcomes, the completeness of data including attrition and exclusions from the analysis. We state whether attrition and exclusions were reported and the numbers included in the analysis at each stage (compared with the total randomised participants), reasons for attrition or exclusion where reported, and whether missing data were balanced across groups or were related to outcomes. Where sufficient information is reported, or was supplied by the trial authors, we re-included missing data in the analyses which we undertook.

We assessed methods as:

• low risk of bias (e.g. no missing outcome data; missing outcome data balanced across groups);

 high risk of bias (e.g. numbers or reasons for missing data imbalanced across groups; 'as treated' analysis done with substantial departure of intervention received from that assigned at randomisation);

• unclear risk of bias.

#### (5) Selective reporting (checking for reporting bias)

We describe for each included study how we investigated the possibility of selective outcome reporting bias and what we found. We assessed the methods as:

• low risk of bias (where it is clear that all of the study's prespecified outcomes and all expected outcomes of interest to the review have been reported);

• high risk of bias (where not all the study's prespecified outcomes have been reported; one or more reported primary outcomes were not prespecified; outcomes of interest are reported incompletely and so cannot be used; study fails to include results of a key outcome that would have been expected to have been reported);

• unclear risk of bias.

# (6) Other bias (checking for bias due to problems not covered by (1) to (5) above)

We describe for each included study any important concerns we have about other possible sources of bias.

We assessed whether each study was free of other problems that could put it at risk of bias:

- low risk of other bias;
- high risk of other bias;
- unclear whether there is risk of other bias.

#### (7) Overall risk of bias

We made explicit judgements about whether studies were at high risk of bias, according to the criteria given in the *Handbook* ( Higgins 2011). With reference to (1) to (6) above, we assessed the likely magnitude and direction of the bias and whether we considered it was likely to impact on the findings. We explored the

impact of the level of bias through undertaking sensitivity analyses - *see* Sensitivity analysis.

#### Measures of treatment effect

#### **Dichotomous data**

For dichotomous data, we presented results as summary risk ratio with 95% confidence intervals.

#### Continuous data

For continuous data, we used the mean difference if outcomes were measured in the same way between trials. We used the standardised mean difference to combine trials that measured the same outcome, but used different methods.

#### Unit of analysis issues

Trials with two treatment groups were analysed by including each treatment arm in separate meta-analyses.

#### Dealing with missing data

For included studies, we noted levels of attrition. We planned to explore the impact of including studies with high levels of missing data in the overall assessment of treatment effect by using sensitivity analysis. For all outcomes, we carried out analyses, as far as possible, on an intention-to-treat basis, i.e. we attempted to include all participants randomised to each group in the analyses, and all participants were analysed in the group to which they were allocated, regardless of whether or not they received the allocated intervention. The denominator for each outcome in each trial was the number randomised minus any participants whose outcomes were known to be missing.

#### Assessment of heterogeneity

We assessed statistical heterogeneity in each meta-analysis using the T<sup>2</sup>, I<sup>2</sup> and Chi<sup>2</sup> statistics. We regarded heterogeneity as substantial if T<sup>2</sup> was greater than zero and either the I<sup>2</sup> was greater than 30% or there was a low P value (less than 0.10) in the Chi<sup>2</sup> test for heterogeneity.

#### Assessment of reporting biases

In future updates, if there are 10 or more studies in the meta-analysis we will investigate reporting biases (such as publication bias) using funnel plots. We will assess funnel plot asymmetry visually, and use formal tests for funnel plot asymmetry. For continuous outcomes we will use the test proposed by Egger 1997, and for dichotomous outcomes we will use the test proposed by Harbord 2006. If asymmetry is detected in any of these tests or is suggested by a visual assessment, we will perform exploratory analyses to investigate it.

#### Data synthesis

We carried out statistical analysis using the Review Manager software (RevMan 2011). We use fixed-effect meta-analysis for combining data where it was reasonable to assume that studies were estimating the same underlying treatment effect: i.e. where trials examined the same intervention, and the trials' populations and methods were judged sufficiently similar. If there was clinical heterogeneity sufficient to expect that the underlying treatment effects differed between trials, or if substantial statistical heterogeneity was detected, we used random-effects meta-analysis to produce an overall summary if an average treatment effect across trials was considered clinically meaningful. The random-effects summary was treated as the average range of possible treatment effects and we discussed the clinical implications of treatment effects differing between trials. If the average treatment effect was not clinically meaningful, we did not combine trials.

Where we use random-effects analyses, the results are presented as the average treatment effect with 95% confidence intervals, and the estimates of  $T^2$  and  $I^2$ .

#### Sensitivity analysis

Sensitivity analysis was not required.

# RESULTS

### **Description of studies**

See: Characteristics of included studies; Characteristics of excluded studies.

Twenty-nine studies met the inclusion criteria for this review but only 22 (1793 women) contributed data. For full details of each trial, *see* Characteristics of included studies.

Nine trials were excluded, for more details, *see* Characteristics of excluded studies. Seven trial reports are in Studies awaiting classification.

### **Included studies**

In 18 of the trials, the indication for caesarean section was nonurgent and the women were healthy and stable. In six of the remaining eleven trials, the indication for caesarean was severe preeclampsia in three, pre-eclampsia with non-reassuring heart trace in one, pre-eclampsia with caesarean section indications other than fetal distress in another, and pregnancy-induced hypertension in

Regional versus general anaesthesia for caesarean section (Review) Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

the sixth one. In the remaining five trials, two were said to have been either emergency or elective, two were emergency only and the last one was said to be for 'proven uteroplacental insufficiency'. Bupivacaine was used for regional anaesthesia in 17 of the trials; other local anaesthetic agents used include lidocaine, levobupivacaine and ropivacaine. For general anaesthesia thiopentone, suxamethonium and a mixture of nitrous oxide and oxygen were used for induction in 15 of the trials. Five trials used thiopentone and succinyl choline. Other combinations used include thiopentone and rocuronium, propofol and succinylcholine and propofol and cis-atracurium. Six of the papers reported the use of halothane as well, six used isoflurane, another six-sevoflurane, one-enflurane and two did not report the use of any volatile agent for anaesthesia.

#### **Risk of bias in included studies**

Details for each trial are in the Characteristics of included studies table. Many of the studies were small and most of them did not report their method of randomisation or allocation concealment. Blinding of outcome assessments was done in some studies. Only one study analysed the data in an intention-to-treat manner. Intervention and control groups were comparable in all but one of the studies, in which this was not reported.

# **Effects of interventions**

### **Maternal outcomes**

Maternal deaths

No trial reported on deaths.

#### Pre and postoperative haematocrit

One study (Lertakyamanee 1999) reported a significant difference which favoured epidural anaesthesia (mean difference (MD) 1.70%; 95% confidence interval (CI) 0.47 to 2.93; 231 women (Analysis 1.13)) and spinal anaesthesia (MD 3.10%; 95% CI 1.73 to 4.47; 209 women (Analysis 2.12)) over general anaesthesia.

#### Maternal blood loss

Three trials reported on estimated maternal blood loss. Compared with general anaesthesia, there was significantly less estimated blood loss when using epidural anaesthesia (standardised mean difference (SMD) -0.32 mL; 95% CI -0.56 to -0.07; two trials (Hong 2002; Lertakyamanee 1999), 256 women (Analysis 1.12)) or spinal anaesthesia (SMD -0.59; 95% CI -0.83 to -0.35; two trials (Lertakyamanee 1999; Dyer 2003), 279 women (Analysis 2.11)).

#### Wound and other infections

No study reported on wound and other infections.

#### Pain

One study (Lertakyamanee 1999) reported the occurrence of intraoperative pain. It reported that the perception of pain during the caesarean section was less when general anaesthesia was used compared with epidural anaesthesia (MD 0.84; 95% CI 0.45 to 1.23, 223 women) or spinal anaesthesia (MD 0.69; 95% CI 0.32 to 1.06, 221 women) (Analysis 1.23 and Analysis 2.21). However, one study (Hong 2002) reported that the time to request for analgesia postoperatively was longer with epidural compared with general anaesthesia (MD 500.00 minutes; 95% CI 364.36 to 635.64, 25 women) (Analysis 1.22). Another study (Turhanoglu 1999) reported that the time to request postoperative analgesia was longer with spinal than with general anaesthesia (MD 97.80 minutes; 95% CI 90.28 to 105.32; 60 women (Analysis 2.19).

#### Satisfaction

One study (Lertakyamanee 1999) reported on satisfaction using a visual analogue score and noted that there was no difference in the level of satisfaction when general anaesthesia was compared with either spinal anaesthesia (MD -0.58; 95% CI -1.26 to 0.10; 221 women (Analysis 2.15)) or epidural anaesthesia (MD -0.01; 95% CI -0.63 to 0.61; 223 women (Analysis 1.14)). However, compared with the epidural or spinal group, more women in the general anaesthesia group stated they would use the same technique again if they needed caesarean section for a subsequent pregnancy (epidural versus general anaesthesia: risk ratio (RR) 0.80; 95% CI 0.65 to 0.98; one trial, 223 women (Analysis 1.15); spinal versus general anaesthesia: RR 0.80; 95% CI 0.65 to 0.99; one trial, 221 women Analysis 2.14)).

#### Adverse events

There was no difference in any of the adverse events reported.

#### Neonatal outcomes

#### **Neonatal deaths**

No study reported on neonatal deaths.

#### Umbilical artery pH

Eight studies reported on the mean umbilical artery pH in mothers who had epidural anaesthesia: (Hollmen 1978; Datta 1983; Dick 1992; Wallace 1995; Pence 2002; Petropoulos 2003; Yegin 2003; Bengi Sener 2003).

Indications for caesarean section were not urgent in seven of the eight trials, and from these trials there was no difference in the pH in babies whose mothers received epidural anaesthesia compared to general anaesthesia (average MD 0.00; 95% CI -0.01 to 0.02; random-effects,  $T^2 = 0.00$ ,  $I^2 = 60\%$ ; seven studies, 397 women (Analysis 1.1)). When including the Wallace 1995 study, in which indication for caesarean section was urgent, there was also no overall difference when all the eight trials were combined (average MD 0.00; 95% CI -0.02 to 0.01; random-effects,  $T^2 = 0.00$ ,  $I^2 = 67\%$ , 454 women (Analysis 1.1)). These studies showed a significant degree of heterogeneity both for those with non-urgent indications for caesarean section ( $I^2 = 60\%$ ), and overall ( $I^2 = 67\%$ ).

Six trials (Datta 1983; Mahajan 1992; Kavak 2001; Akyol 2006; Moslemi 2007; Mancuso 2010) also reported no significant difference in the mean umbilical artery pH when mothers had received spinal anaesthesia compared to general anaesthesia (MD -0.00; 95% CI -0.01 to 0.00; 459 women (Analysis 2.1)). Dyer 2003 reported a lower median umbilical artery pH when mothers had received spinal compared to general anaesthesia (66 women, *see* Table 1).

Where both spinal and epidural anaesthesia were given in the same woman and compared with general anaesthesia, two studies (Wallace 1995; Petropoulos 2003) found the mean umbilical artery pH to be significantly lower compared with the general anaesthesia group (MD -0.03; 95% CI -0.04 to -0.02; 211 women; (Analysis 3.1)).

#### Umbilical vein pH

Seven studies (Hollmen 1978; Datta 1983; Dick 1992; Mahajan 1992; Lertakyamanee 1999; Yegin 2003; Yentur 2009) reported on the mean umbilical vein pH in mothers who had epidural anaesthesia. There was no significant difference in pH in babies whose mothers had received epidural anaesthesia compared with general anaesthesia (average MD 0.01; 95% CI -0.01 to 0.02; random-effects,  $T^2 = 0.00$ ,  $I^2 = 59\%$ , 505 women (Analysis 1.2)). Four trials (Datta 1983; Mahajan 1992; Lertakyamanee 1999; Nabhan 2009) found that the mean umbilical vein pH of children whose mothers had received spinal anaesthesia was higher than those whose mothers had received spinal anaesthesia, but this difference did not reach statistical significance either (MD 0.01; 95% CI 0.00 to 0.02; 383 women; (Analysis 2.2)).

#### Neonatal neurological adaptive score

Two studies (Lertakyamanee 1999; Bengi Sener 2003) reported on the mean adaptive score at two to four hours and noted that there were no differences in the scores in babies delivered following general anaesthesia when compared with epidural anaesthesia (average MD 2.17; 95% CI -1.13 to 5.47; random-effects,  $T^2 =$ 5.38,  $I^2 = 95\%$ ; 253 women (Analysis 1.9)).

When looking at the proportion of babies with neurologic and adaptive capacity scores less than 35, Mahajan 1992 noted that

there were no differences in the epidural group when compared with the general anaesthesia group at 15 minutes (RR 0.94; 95% CI 0.62 to 1.45, 60 women (Analysis 1.16)) and at two hours (RR 0.67; 95% CI 0.27 to 1.64, 60 women (Analysis 1.17)).

One study (Lertakyamanee 1999) documented the mean adaptive score at two to four hours and noted that there were also no differences in babies whose mothers had received spinal anaesthesia over general anaesthesia (MD 0.40; 95% CI -0.54 to 1.34; 221 women (Analysis 2.3)). In contrast, Mahajan 1992 reported significantly fewer children with adaptive scores less than 35 were born to women who received spinal anaesthesia compared with general anaesthesia at 15 minutes (RR 0.17; 95% CI 0.05 to 0.51; 60 women (Analysis 2.4)) and at two hours (RR 0.05; 95% CI 0.00 to 0.87; 60 women (Analysis 2.5)).

#### Apgar score

Five studies (Hodgkinson 1980; Lertakyamanee 1999; Mathur 2002; Yegin 2003; Yentur 2009) documented mean Apgar score at one minute comparing epidural anaesthesia with general anaesthesia. They reported that there was no significant difference between the two groups (average MD 0.39; 95% CI -0.34 to 1.12; random-effects,  $T^2 = 0.52$ ,  $I^2 = 87\%$ ; 408 women (Analysis 1.7)). After analysing five studies comparing spinal with general anaesthesia, (Lertakyamanee 1999; Kavak 2001; Mathur 2002; Akyol 2006; Moslemi 2007), we again found no significant difference in mean Apgar score at one minute (average MD 0.54; 95% CI - 0.09 to 1.16; random-effects,  $T^2 = 0.43$ ,  $I^2 = 90\%$ ; 470 women (Analysis 2.7)).

Four studies comparing epidural with general anaesthesia reported Apgar scores at five minutes (Hodgkinson 1980; Lertakyamanee 1999; Yegin 2003; Yentur 2009) and did not find any significant difference between the two groups (average MD 0.20; 95% CI -0.15 to 0.55; random-effects,  $T^2 = 0.08$ ,  $I^2 = 72\%$ , 368 women (Analysis 1.8)). Furthermore, when comparing spinal with general anaesthesia, four studies (Lertakyamanee 1999; Kavak 2001; Akyol 2006; Moslemi 2007) did not find any significant differences in the mean Apgar score at five minutes (average MD 0.24; 95% CI -0.05 to 0.54; random-effects,  $T^2 = 0.07$ ,  $I^2 = 83\%$ ; 429 women (Analysis 2.8)).

One study (Korkmaz 2004) also found no significant differences in mean Apgar score at one minute (MD 0.25; 95% CI -0.14 to 0.64; 30 women (Analysis 3.5)) or at five minutes (not estimable), when comparing combined spinal and epidural anaesthesia with general anaesthesia. Mathur et al (Mathur 2002) did not find any significant differences in mean Apgar score at ten minutes either, when comparing spinal with general anaesthesia (MD -0.15; 95% CI -0.81 to 0.51, 41 women (Analysis 2.18)), or when comparing epidural with general anaesthesia (MD -0.45; 95% CI -1.03 to 0.13; 40 women (Analysis 1.21)).

One study (Dick 1992) reported on the proportion of babies with Apgar score of four or less, comparing epidural with general anaes-

Regional versus general anaesthesia for caesarean section (Review)

Copyright  $\ensuremath{\textcircled{0}}$  2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

thesia. The trial did not find any significant difference in the proportion of children with such low Apgar scores at one minute (RR 0.15; 95% CI 0.01 to 2.73; 47 women (Analysis 1.3)) and at five minutes (RR 0.35; 95% CI 0.01 to 8.11; 47 women (Analysis 1.4)).

Two studies (Wallace 1995; Petropoulos 2003) reported on the proportion of babies with Apgar scores six or less in women who received epidural versus general anaesthesia. No significant difference was found at one minute (RR 0.58; 95% CI 0.22 to 1.52; 209 women (Analysis 1.5)) or at five minutes (RR 0.46; 95% CI 0.11 to 1.95; 209 women (Analysis 1.6)). In comparing an Apgar score of six or less at one minute in women who received spinal versus general anaesthesia, five studies (Mahajan 1992; Turhanoglu 1999; Moslemi 2007; Nabhan 2009; Mancuso 2010) found significantly fewer babies in the spinal group (average RR 0.54; 95% CI 0.16 to 1.84; random-effects, T<sup>2</sup> = 1.32, I<sup>2</sup> = 77%, 435 women (Analysis 2.9)). At five minutes however, three studies (Mahajan 1992; Turhanoglu 1999; Mancuso 2010) did not find significant differences between groups in terms of Apgar score of six or less (RR 0.56; 95% CI 0.21 to 1.46; 291 women (Analysis 2.10)). No significant differences were seen in the proportion of babies with an Apgar score of six or less when combined epidural and spinal anaesthesia were used compared with general anaesthesia at one minute (RR 0.89; 95% CI 0.38 to 2.07; two trials, 211 women (Analysis 3.2)) and at five minutes (RR 0.60; 95% CI 0.15 to 2.44; 211 women (Analysis 3.3)) in two studies (Wallace 1995; Petropoulos 2003).

#### Need for oxygen for resuscitation

One study (Petropoulos 2003) comparing epidural versus general anaesthesia reported on the need for oxygen and did not find a significant difference in the need for oxygen (RR 0.86; 95% CI 0.34 to 2.20; 152 women (Analysis 1.19)). Similarly, when combined epidural and spinal anaesthesia were compared with general anaesthesia (Petropoulos 2003), no significant difference was observed for this outcome (RR 1.14; 95% CI 0.49 to 2.65; 158 women (Analysis 3.4)).

# DISCUSSION

No trial reported on maternal or neonatal deaths. This attests to the relative safety of caesarean section, especially in the countries where most of the trials in this review were conducted. Larger sample sizes would be needed to detect such outcomes but many of the included studies were underpowered. Also, as a result of the massive shift to the use of regional anaesthesia for caesarean section for maternal reasons (Hawkins 1997b), it is unlikely that such studies will be performed in future.

From the results, epidural and spinal anaesthesia appears to be associated with less blood loss than general anaesthesia. Although this did not translate into a reduction in the need for blood transfusion, it may be clinically significant especially as anaemia is detrimental to postoperative wellbeing and healing.

The finding of less intraoperative pain with general anaesthesia compared with both forms of regional is not surprising as this method is characterised by the abolishment of all sensation. The time to request analgesia, an index of postoperative pain requirements, appeared longer when mothers were given epidural anaesthesia. This is probably because drugs administered into the epidural space last longer and the epidural catheter can be left in situ and topped up for several hours after surgery. However, there was a big difference in standard deviation between the epidural and general anaesthesia groups. This suggests differing distribution of data and some degree of skewing, rendering the mean difference comparisons invalid. Thus, firm conclusions cannot be drawn from these data. With spinal anaesthesia, the longer time to request analgesia was also found in one study. Despite the relatively small sample size, this is an expected advantage of regional anaesthesia as its anaesthetic effect usually persists after the surgical procedure.

One of the big issues in healthcare delivery is client satisfaction; only one trial reported on this and did not find any differences in satisfaction between regional and general anaesthesia. It is clear that this aspect would need to be addressed in the design of new trials. In terms of preference of the same technique again, however, based on the results of one study, women who had general anaesthesia appeared to favour it over regional anaesthesia (both spinal and epidural) for caesarean sections. The reasons for this preference could not be determined from the study.

Regarding neonatal outcomes, neither umbilical artery or vein pH was affected by spinal or epidural anaesthesia when the indications for surgery are not urgent. This differs from the findings of a recent meta-analysis (Reynolds 2005) that showed that spinal anaesthesia resulted in lower umbilical cord pH results than general, but showed no difference when epidural anaesthesia was compared with general anaesthesia. The authors however included both randomised and non-randomised trials and combined both umbilical artery and vein pH data in their analysis of cord pH. Umbilical blood sampling is one of the parameters used in defining and deciding how aggressively one should resuscitate any baby with severe birth asphyxia. Although umbilical artery pH appeared to be favoured by the use of general as compared with combined spinal and epidural anaesthesia, the differences found in this review may not be clinically significant as the mean figures were within normal neonatal limits (7.11 to 7.45). They were also well above the cutoff for defining acidosis (pH less than 7.0) (Stoll 2000).

The Apgar score is a composite measure of the clinical and cardiorespiratory status of the baby at birth. It is measured usually at one minute (to determine the extent of resuscitation required) and at five minutes (to determine the response to resuscitation and to diagnose asphyxia). Apart from a higher mean Apgar score at

one minute in epidural versus general anaesthesia groups, there were no other differences in Apgar score measurements between the three intervention comparison groups. We can thus conclude that practically, one form of anaesthesia has not been shown to be superior to the other, as far as the determination of asphyxia is concerned.

The neonatal neurological adaptive score is an attempt to measure the neurological status of the babies on the assumption that the drugs used in inducing anaesthesia may depress the central nervous system. Overall, the results suggest that regional anaesthesia conveys a more favourable outcome than general anaesthesia, especially when babies are categorised based on a cut-off point of less than 35 and particularly when spinal anaesthesia is used. However, the study which showed a significant effect with spinal anaesthesia was a small one with 30 participants in each arm (Mahajan 1992). Also, there was significant heterogeneity between the two studies that showed a difference in the mean score within two to four hours using epidural anaesthesia (Lertakyamanee 1999; Bengi Sener 2003), which could be because the former study (30 women) was much smaller than the latter (223 women). The larger study did not show a significant difference in the mean score between epidural and general anaesthesia.

# AUTHORS' CONCLUSIONS

# References to studies included in this review

#### Akyol 2006 {published data only}

Akyol A, Akgun A, Gedikbasi A, Agrali G, Ceylan Y. Effects of general and spinal anesthesia on APGAR scores and umblical cord blood gases in elective cesarean operations [Elektif sezaryen operasyonlarinda genel ve spinal anestezinin yenidogan APGAR puanina ve kordon kan gazlarina etkisi]. *Jinekoloji ve Obstetrik Dergisi* 2006;**20** (1):32–7.

#### Bengi Sener 2003 {published data only}

Bengi Sener E, Guldogus F, Karakaya D, Baris S, Kocamanoglu, Tur A. Comparison of neonatal effects of epidural and general anesthesia for cesarean section. *Gynecologic and Obstetric Investigation* 2003;**55**:41–5.

#### Braithwaite 1993 {published data only}

Braithwaite H. Comparative study of time taken to produce surgical readiness between spinal anaesthesia and general anaesthesia for emergency caesarean section. *South African Journal of Family Practice* 1993;**14**(2):46–9.

#### Implications for practice

There is not enough evidence from this review to show that either regional or general anaesthesia is superior to the other in terms of major maternal or neonatal outcomes. Thus, the choice of one over the other lies with other criteria such as estimated blood loss which appears to be reduced with the use of regional anaesthesia. This may assume greater importance depending on the context in which one is operating.

#### Implications for research

Trials measuring outcomes such as maternal and newborn morbidity, maternal satisfaction with techniques and adverse events are necessary.

# ACKNOWLEDGEMENTS

We are grateful to Michael Mestre who translated two of the papers into English from Turkish and Lynn Hampson for updating the search and finding a translator.

Nkihu Merah for her contribution to earlier versions of this review.

As part of the pre-publication editorial process, this review has been commented on by three peers (an editor and two referees who are external to the editorial team), a member of the Pregnancy and Childbirth Group's international panel of consumers and the Group's Statistical Adviser.

# REFERENCES

#### Datta 1983 {published data only}

Datta S, Carr DB, Lambert DH, Morrison J, Naulty JS, Fischer J, et al.Anesthesia for cesarean delivery: relationship of maternal and fetal plasma B-endorphin concentrations to different types of anesthesia. *Anesthesiology* 1983;**59**:A418.

### Dermitzaki 2009 {published data only}

Dermitzaki E, Staikou C, Petropoulos G, Rizos D, Siafaka I, Fassoulaki A. A randomized study of maternal serum cytokine levels following cesarean section under general or neuraxial anesthesia. *International Journal of Obstetric Anesthesia* 2009;**18**(1):33–7.

#### Dick 1992 {published data only}

\* Dick W, Traub E, Kraus H, Tollner U, Burghard R, Muck J. General anaesthesia versus epidural anaesthesia for primary caesarean section-a comparative study. *European Journal of Anaesthesiology* 1992;**9**:15–21. Dick W, Traub E, Kraus H, Tollner U, Burghardt R, Muck J. General anaesthesia vs epidural anaesthesia for primary caesarean section - a comparative study. *Geburtshilfe und Frauenheilkunde* 1993;**53**:595.

#### Dogan 2008 {published data only}

Dogan R, Birdane A, Bilir A, Ekemen S, Tanriverdi B. Frequency of electrocardiographic changes indicating myocardial ischemia during elective cesarean delivery with regional and general anesthesia: detection based on continuous holter monitoring and serum markers of ischemia. *Journal of Clinical Anesthesia* 2008;**20**(5):347–51.

#### Dyer 2003 {published data only}

Dyer R, Els I, Farbas J, Schoeman L, Torr G, James M. A randomised trial comparing general with spinal anaesthesia for caesarean section in preeclamptics with a non-reassuring fetal heart trace [abstract]. *International Journal of Obstetric Anesthesia* 2003;**12**:202.

\* Dyer RA, Els I, Farbas J, Torr GJ, Schoeman LK, James MF. Prospective, randomized trial comparing general with spinal anesthesia for cesarean delivery in preeclamptic patients with a nonreassuring fetal heart trace. *Anesthesiology* 2003;**99**:561–9.

#### Hodgkinson 1980 {published data only}

Hodgkinson R, Hussain FJ, Hayashi RH. Systemic and pulmonary blood pressure during caesarean section in parturients with gestational hypertension. *Canadian Anaesthetists Society Journal* 1980;**27**:389–94.

#### Hollmen 1978 {published data only}

Hollmen AI, Jouppila R, Koivisto M, Maatta L, Pihlajaniemi R, Puukka M, et al.Neurologic activity of infants following anesthesia for cesarean section. *Anesthesiology* 1978;**48**: 350–6.

#### Hong 2002 {published data only}

Hong JY, Jee YS, Yoon HJ, Kim SM. Comparison of general and epidural anesthesia in elective cesarean section for placenta previa totalis: maternal hemodynamics, blood loss and neonatal outcome. *International Journal of Obstetric Anesthesia* 2003;**12**:12–6.

#### Jain 2009 {published data only}

Jain K, Sharma A, Bhardwaj N, Kalra J. General or low dose in sa in patients with growth restricted festuses for cesarean delivery. American Society of Anaesthesiologists Annual Meeting; 2009 Oct 17-21; New Orleans, USA. 2009.

#### Kavak 2001 {published data only}

Kavak ZN, Basgul A, Ceyhan N. Short-term outcome of newborn infants: spinal versus general anesthesia for elective cesarean section. A prospective randomized study. *European Journal of Obstetrics & Gynecology and Reproductive Biology* 2001;**100**:50–4.

#### Kim 2000 {published data only}

Kim DY, Han JI. The effects of anesthetic techniques and postoperative pain control on the response of the stress and immune system in cesarean section patients. *Korean Journal* of Anesthesiology 2000;**39**(5):673–8.

### Korkmaz 2004 {published data only}

Korkmaz F, Eksioglu B, Hanci A, Basgul A. Comparison of combined spinal epidural block and general anesthesia for cesarean section [abstract]. *Regional Anesthesia and Pain Medicine* 2004;**29**(Suppl 2):77.

#### Lertakyamanee 1999 {published data only}

Kolatat T, Somboonnanonda A, Lertakyamanee J, Chinachot T, Tritrakarn T, Muangkasem J. Effects of general and regional anesthesia on the neonate (a prospective, randomized trial). *Journal of the Medical Association of Thailand* 1999;**82**:40–5.

\* Lertakyamanee J, Chinachoti T, Tritrakarn T, Muangkasem J, Somboonnanonda A, Kolatat T. Comparison of general and regional anesthesia for cesarean section: success rate, blood loss and satisfaction from a randomized trial. *Journal of the Medical Association of Thailand* 1999;**82**:672–80.

#### Mahajan 1992 {published data only}

Mahajan J, Mahajan RP, Singh MM, Anand NK. Anaesthetic technique for elective caesarean section and neurobehavioural status of newborns. *International Journal* of Obstetric Anesthesia 1992;**2**:89–93.

#### Mancuso 2010 {published data only}

Mancuso A, De Vivo A, Giacobbe A, Priola V, Savasta LM, Guzzo M, et al.General versus spinal anaesthesia for elective caesarean sections: effects on neonatal short-term outcome. A prospective randomised study. *Journal of Maternal-Fetal* & Neonatal Medicine 2010;**23**(10):1114–8.

# Mathur 2002 {published data only}

Mathur V, Nagrath A, Saxena S, Maheshwari M. Study of the maternal and neonatal outcome in caesarean deliveries in pregnancies complicated by pre-eclampsia conducted under general anaesthesia and regional anaesthesia. *Journal* of Obstetrics and Gynecology of India 2002;**52**(4):28–32.

#### Momani 2001 {published data only}

Momani O. Controlled trial of wound infiltration with bupivacaine for post operative pain relief after caesarean section. *Bahrain Medical Bulletin* 2001;**23**(2):84–6.

#### Moslemi 2007 {published data only}

Moslemi F, Rasooli S. Comparison of spinal versus general anesthesia for cesarean delivery in patients with severe preeclampsia. *Journal of Medical Sciences* 2007;7(6):1044–8.

### Nabhan 2009 {published data only}

Nabhan AF, El-Din LB, Rabie AH, Fahmy GM. Impact of intrapartum factors on oxidative stress in newborns. *Journal of Maternal-Fetal & Neonatal Medicine* 2009;**22** (10):867–72.

#### Papadopoulou 2005 {published data only}

Papadopoulou E. Maternal stress response during emergency cesarean section with general and spinal anesthesia. *Regional Anesthesia and Pain Medicine* 2005;**30** (Suppl 1):72.

#### Pence 2002 {published data only}

Pence S, Kocoglu H, Balat O, Balat A. The effect of delivery on umbilical arterial cord blood gases and lipid peroxides: comparison of vaginal delivery and cesarean section. *Clinical and Experimental Obstetrics and Gynecology* 2002;**29**:212–4.

#### Petropoulos 2003 {published data only}

Petropoulus G, Siristatidis C, Salamalekis E, Creatsas G. Spinal and epidural versus general anesthesia for elective cesarean section at term: effect on the acid-base status of the

Regional versus general anaesthesia for caesarean section (Review)

mother and newborn. *Journal of Maternal-Fetal & Neonatal Medicine* 2003;**13**:260–6.

### Turhanoglu 1999 {published data only}

Turhanoglu S, Tok D, Ozyilmaz MA, Bayhan N, Olmez G, Kaya S, et al.Comparison of general and spinal anaesthesia in pregnant women with pregnancy induced hypertension [Gebeligin olusturugu hipertansiyonlu olgularda genel ve spinal anestezinen karsilastirilmasi]. *Turk Anesteziyoloji Ve Reanimasyon* 1999;**27**(9):480–5.

# Wallace 1995 {published data only}

\* Wallace DH, Leveno KJ, Cunningham FG, Giesecke AH, Shearer VE, Sidawi JE. Randomized comparison of general and regional anesthesia for cesarean delivery in pregnancies complicated by severe preeclampsia. *Obstetrics & Gynecology* 1995;**86**:193–9.

Wallace DH, Leveno KJ, Cunningham FG, Shearer V, Black S, Holloway J. Randomized study of general anesthesia vs epidural or spinal-epidural analgesia for cesarean section in pregnancies complicated by severe preeclampsia. *American Journal of Obstetrics and Gynecology* 1992;**166**:302.

#### Yegin 2003 {published data only}

Yegin A, Ertug Z, Yilmaz M, Erman M. The effects of epidural anesthesia and general anesthesia on newborns at cesarean section. *Turkish Journal of Medical Sciences* 2003; **33**:311–4.

#### Yentur 2009 {published data only}

Yentur EA, Topcu I, Ekici Z, Ozturk T, Keles GT, Civi M. The effect of epidural and general anesthesia on newborn rectal temperature at elective cesarean section. *Brazilian Journal of Medical & Biological Research* 2009;**42**(9):863–7.

#### References to studies excluded from this review

#### Abboud 1985 {published data only}

Abboud TK, Nagappala S, Murakawa K, David S, Haroutunian S, Zakarian M, et al.Comparison of the effects of general and regional anesthesia for cesarean section on neonatal neurologic and adaptive capacity scores. *Anesthesia* & *Analgesia* 1985;**64**:996–1000.

#### Akturk 1995 {published data only}

Akturk G, Kiral N, Barlak A, Solak M, Ozen I, Erciyes N. The choice of anaesthetic technique for caesarean section does not affect plasma B-endorphin levels in the neonate. *European Journal of Anaesthesiology* 1995;**12**:525–7.

#### Fyneface-Ogan 2008 {published data only}

Fyneface-Ogan S, Uzoigwe SA. Caesarean section outcome in eclamptic patients: a comparison of infiltration and general anaesthesia. *West African Journal of Medicine* 2008; **27**(4):250–4.

#### Gambling 1995 {published data only}

Gambling DR, Sharma SK, White PF, Beveren TV, Bala AS, Gouldson R. Use of sevoflurane during elective cesarean birth: a comparison with isoflurane and spinal anesthesia. *Anesthesia & Analgesia* 1995;**81**:90–5.

#### Kamat 1991 {published data only}

Kamat SK, Shah MV, Chaudhary LS, Pandya S, Bhatt MM. Effect of induction - delivery and uterine delivery on apgar scoring of the newborn. *Journal of Postgraduate Medicine* 1991;**37**(3):125–7.

#### Navarro 2000 {published data only}

Navarro EM. Desflurane-general anesthesia for cesarean section compared with isoflurane and epidural anesthesia [Desfluran–Allgemeinanasthesie zur Sectio caesarea: Vergleich mit Isofluran und Epiduralanasthesie]. *Anasthesiologie, Intensivmedizin, Notfallmedizin, Schmerztherapie* 2000;**35**:232–6.

#### Qublan 2001 {published data only}

Qublan HS, Merhej A, Dabbas MA, Hindawi IM. Spinal versus general anesthesia for elective cesarean delivery: a prospective comparative study. *Clinical and Experimental Obstetrics and Gynecology* 2001;**28**:246–8.

#### Ratcliffe 1992 {published data only}

Ratcliffe FM, Evans JM. Neonatal wellbeing after elective caesarean delivery with general, spinal, and epidural anaesthesia. *European Journal of Anaesthesiology* 1993;**10**: 175–81.

#### White 1962 {published data only}

White CW, Weiss JB, Alver EC, Heerdegen DK. Anesthesia and postpartum headache. *Obstetrics & Gynecology* 1962; **20**:734–8.

# References to studies awaiting assessment

#### Dasgupta 2011 {published data only}

Dasgupta S, Chakraborty B, Saha D, Ghosh D. Comparison of neonatal outcome in women with severe pre-eclampsia undergoing caesarean section under spinal or general anaesthesia. *Journal of the Indian Medical Association* 2011; **109**(3):166–70.

#### Demiraran 2011 {published data only}

Demiraran Y, Albayrak M, Seker IS, Kaynak G, Iskender A, Sezen GY, et al.Effect of anesthesiological strategies on neonatal bilirubin levels during cesarean section: a prospective and randomized trial. *Archives of Gynecology and Obstetrics* 2011;**284**(5):1059–65.

#### Geze 2011 {published data only}

Geze S, Guven S, Cekic B, Karahan C, Erturk E, Mentese A. Effects of spinal and general anesthesia on umbilical cord malondialdehydes and ischemia modified albumin levels in cesarean section. *Regional Anesthesia and Pain Medicine* 2011;**36**(5 Suppl 2):E162.

#### Hawwas 2011 {published data only}

Hawwas F, Acar U, Orhan Sungur M, Karadeniz M, Ozkan Seyhan T. Spinal versus general anesthesia for maternal and fetal outcomes in elective cesarean section. *Regional Anesthesia and Pain Medicine* 2011;**36**(5 Suppl 2):E208.

#### Saracoglu 2012 {published data only}

Saracoglu KT, Saracoglu A, Umuroglu T, Eti Z. Neuraxial block versus general anaesthesia for cesarean section: Post-operative pain scores and analgesic requirements. *Journal of the Pakistan Medical Association* 2012;**62**(5):441–4.

#### Waris 2002 {published data only}

Waris S, Yousuf M, Ahmed RA, Shahid M. An experience of spinal anesthesia versus general anesthesia in severe pre-

Regional versus general anaesthesia for caesarean section (Review)

eclamptic patients undergoing lower segment cesarean section. *Journal of Surgery Pakistan* 2002;7(2):25–7.

#### Wei 2009 {published data only}

Wei J, Liu GL, Liang MY, Wang SM. [Effect of general anesthesia used in cesarean section on maternalneonatal outcome of pregnancy complicated with severe thrombocytopenia]. [Chinese]. Chung-Hua Fu Chan Ko Tsa Chih [Chinese Journal of Obstetrics & Gynecology] 2009; 44(9):665–8.

### Additional references

# Ajmal 2011

Ajmal M. General anaesthesia for caesarean sections: are anaesthetists dealing with exaggerated fear?. *European Journal of Anaesthesiology* 2011;**28**:815–6.

# Andersen 1987

Andersen HF, Auster GH, Marx GF, Merkatz IR. Neonatal status in relation to incision intervals, obstetric factors, and anesthesia at cesarean delivery. *American Journal of Perinatology* 1987;**4**:279–83.

#### Andrews 1992

Andrews WW, Ramin SM, Maberry MC, Shearer V, Black S, Wallace DH. Effect of type of anesthesia on blood loss at elective repeat cesarean section. *American Journal of Perinatology* 1992;**9**:197–200.

#### Danforth 1985

Danforth DN. Caesarean section. JAMA 1985;253:811-8.

#### Djabatey 2009

Djabatey EA, Barclay PM. Difficult and failed intubation in 3430 obstetric general anaesthetics. *Anaesthesia* 2009;**64**: 1168–71.

### Egger 1997

Egger M, Davey Smith G, Schneider M, Minder C. Bias in meta-analysis detected by a simple, graphical test. *BMJ* 1997;**315**(7109):629–34.

#### Enkin 2000

Enkin M, Keirse MJNC, Neilson J, Crowther C, Duley L, Hodnett E, et al.*A Guide to Effective Care in Pregnancy and Childbirth.* 3rd Edition. New York: Oxford University Press, 2000.

#### Fox 1979

Fox GS, Smith JB, Namba Y, Johnson RC. Anesthesia for cesarean section: further studies. *American Journal of Obstetrics and Gynecology* 1979;**133**:15–9.

### Furmanik 2010

Furmanik J. A survey of anaesthesia for caesarean section in Poland. *Anestezjologia intensywna terapia* 2010;**42**(2):65–9.

#### Gibbs 1986

Gibbs CP, Krischer J, Peckham BM, Sharp H, Kirschbaum TH. Obstetric anesthesia: a national survey. *Anesthesiology* 1986;**65**:298–306.

#### Harbord 2006

Harbord RM, Egger M, Sterne JA. A modified test for small-study effects in meta-analyses of controlled trials with binary endpoints. *Statistics in Medicine* 2006;**25**(20): 3443–57.

#### Hawkins 1997a

Hawkins JL, Gibbs CP, Orleans M, Martin-Salvaj G, Beaty B. Obstetric anesthesia work force survey, 1981 versus 1992. *Anesthesiology* 1997;**87**:135–43.

#### Hawkins 1997b

Hawkins JL, Koonin LM, Palmer SK, Gibbs CP. Anesthesiarelated deaths during obstetric delivery in the United States, 1979-1990. *Anesthesiology* 1997;**86**:277–84.

#### Hibbard 1996

Hibbard BM, Anderson MM, Drife JO, Tighe JR, Gordon G, Willats S, et al.*Report on Confidential Enquiries into Maternal Deaths in the United Kingdom 1991-1993.* London: HMSO, 1996.

#### Higgins 2011

Higgins JPT, Green S, editors. Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration, 2011. Available from www.cochrane-handbook.org.

#### Juni 2001

Juni P, Altman DG, Egger M. Assessing the quality of controlled clinical trials. *BMJ* 2001;**323**:42–6.

#### Kestin 1991

Kestin IG. Spinal anaesthesia in obstetrics. *British Journal of Anaesthesia* 1991;**66**:596–607.

#### Kolatat 1999

Kolatat T, Somboonnanonda A, Lertakyamanee J, Chinachot T, Tritrakarn T, Muangkasem J. Effects of general and regional anesthesia on the neonate (a prospective, randomized trial). *Journal of the Medical Association of Thailand* 1999;**82**:40–5.

#### Kolås 2006

Kolås T, Saugstad OD, Daltveit AK, Nilsen ST, Øian P. Planned cesarean versus planned vaginal delivery at term: comparison of newborn infant outcomes. *American Journal* of Obstetrics and Gynecology 2006;**195**(6):1538–43.

#### Liu 2007

Liu S, Liston RM, Joseph KS, Heaman M, Sauve R, Kramer MS. Maternal mortality and severe morbidity associated with low-risk planned cesarean delivery versus planned vaginal delivery at term. *Canadian Medical Association Journal* 2007;**176**(4):455–60.

#### Macarthur 2007

Macarthur A, Riley ET. Obstetric anesthesia controversies: vasopressor choice for postspinal hypotension during cesarean delivery. *International Anesthesiology Clinics* 2007; **45**(1):115–32.

#### Ng 2004

Ng KW, Parsons J, Cyna AM, Middleton P. Spinal versus epidural anaesthesia for caesarean section. *Cochrane Database of Systematic Reviews* 2004, Issue 2. [DOI: 10.1002/14651858.CD003765.pub2]

Regional versus general anaesthesia for caesarean section (Review)

#### Ong 1989

Ong BY, Cohen MM, Palahniuk RJ. Anesthesia for cesarean section-effects on neonates. *Anesthesia & Analgesia* 1989; **68**:270–5.

#### RevMan 2003

The Cochrane Collaboration. Review Manager (RevMan). 4.2 for Windows. Oxford, England: The Cochrane Collaboration, 2003.

#### RevMan 2011

The Nordic Cochrane Centre, The Cochrane Collaboration. Review Manager (RevMan). 5.1. Copenhagen: The Nordic Cochrane Centre, The Cochrane Collaboration, 2011.

#### **Reynolds 2005**

Reynolds F, Seed PT. Anaesthesia for caesarean section and neonatal acid-base status: a meta-analysis. *Anaesthesia* 2005; **60**:636–53.

#### Stamer 2005

Stamer UM, Wiese R, Stuber F, Wulf H, Meuser T. Change in anaesthetic practice for Caesarean section in Germany. *Acta Anaesthesiologica Scandinavica 2005* 2005;**49**:170–6.

# Stoll 2000

Stoll BJ, Kliegman RM. The foetus and the neonatal infant.In: Behrman RE, Kleigman RM, Jenson HB editor(s).*Nelson Textbook of Pediatrics*. 16th Edition. Philadelphia:WB Saunders Company, 2000.

#### Thorburn 1998

Thorburn J. Obstetric anaesthesia and analgesia. In: Aitkenhead AR, Smith G editor(s). *Textbook of Anaesthesia*. 3rd Edition. London: Churchill Livingstone, 1998: 533–50.

#### Thorp 2009

Thorp JM. Clinical aspects of normal and abnormal labor. In: Creasy RK, Resnik R, Iams JD, Lockwood CJ, Moore TR editor(s). *Creasy and Resnik's Maternal-Fetal Medicine: Principles and Practice.* 6th Edition. Philadelphia: Saunders Elsevier, 2009:691–724.

### Visalyaputra 2005

Visalyaputra S, Rodanant O, Somboonviboon W, Tantivitayatan K, Thienthong S, Saengchote W. Spinal versus epidural anesthesia for cesarean delivery in severe preeclampsia: a prospective randomized multicenter study. *Anesthesia and Analgesia* 2005;**101**(3):862–8.

# Zagorzycki 1982

Zagorzycki MT, Brinkman CR. The effect of general and epidural anesthesia upon neonatal Apgar scores in repeat cesarean section. *Surgery, Gynecology and Obstetrics* 1982; **155**:641–5.

#### Zanardo 2004

Zanardo V, Simbi AK, Franzoi M, Soldà G, Salvadori A, Trevisanuto D. Neonatal respiratory morbidity risk and mode of delivery at term: influence of timing of elective caesarean delivery. *Acta Paediatrica* 2004;**93**(5):643–7.

#### References to other published versions of this review

#### Afolabi 2006

Afolabi BB, Lesi AFE, Merah NA. Regional versus general anaesthesia for caesarean section. *Cochrane Database* of Systematic Reviews 2006, Issue 4. [DOI: 10.1002/ 14651858.CD004350.pub2]

\* Indicates the major publication for the study

# CHARACTERISTICS OF STUDIES

# Characteristics of included studies [ordered by study ID]

# Akyol 2006

Methods	Women said to have been randomly divided into groups. Blinding of intervention: not done. Blinding of outcome assessment: not stated. No women were excluded from this study. Intention-to-treat analysis: not stated but women remained in their allocated groups
Participants	62 women (30 received spinal, 32 general anaesthesia). Inclusion criteria: ASA-I (American Society of Anesthesiologists) risk group. Women greater than 37 weeks' pregnant, undergoing elective caesarean section. Exclusion criteria: women with complications such as pre-eclampsia, diabetes mellitus, anaemia, premature membrane rupture of fetal complications. Also those with a con- traindication to spinal anaesthesia. Setting: Turkey.
Interventions	Regional group had spinal anaesthesia with bupivacaine. General anaesthesia group had thiopental, rocuronium, and a mixture of nitrous oxide and oxygen and sevoflurane
Outcomes	Review outcomes were mean umbilical arterial pH and Apgar score at 1 and 5 minutes Other study outcomes were umbilical blood gas data such as partial pressures of carbon dioxide and oxygen, as well as bicarbonate concentration

Notes

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

# Bengi Sener 2003

Methods	Women said to have been randomly divided into groups but method not stated. Blinding of intervention: not stated. Blinding of outcome assessment: maternal and neonatal assessments were made by blinded observers. No women were excluded from this study. Intention-to-treat analysis: not stated but women remained in their allocated groups
Participants	30 women (15 received epidural, 15 general anaesthesia). Inclusion criteria: ASA I/II women undergoing elective caesarean section for breech presentation, CPD and previous caesarean section, who had not used regional anaesthesia or analgesia before the study. Exclusion criteria: women with pre-eclampsia, eclampsia, morbid obesity, diabetes mel- litus, anaemia, fetal anomaly, heart disease, marked airway problems, fetal distress, ges- tational age below 37 weeks. Setting: Turkey, University Hospital.
Interventions	Regional group had epidural anaesthesia with bupivacaine. General anaesthesia group had thiopental, succinyl choline, mixture of nitrous oxide and oxygen, isoflurane and vecuronium
Outcomes	Review measured neonatal NACS, umbilical arterial pH, Apgar scores, and adverse events Study outcomes were maternal systolic arterial pressure, heart rate, peripheral oxygen saturation, uterine I-D interval, Apgar scores, neonatal NACS, umbilical arterial blood gases, first breastfeeding interval and complications such as nausea, vomiting and allergic reactions
Notes	

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Braithwaite 1993

Methods	Women said to have been randomly divided into groups by drawing numbers from a hat. Blinding of intervention: not stated. Blinding of outcome assessment: outcome assessments were made by blinded observers. No women were excluded from this study. Intention-to-treat analysis: not stated but women remained in their allocated groups		
Participants	50 women (25 received spinal, 25 general anaesthesia) Inclusion criteria: women undergoing emergency caesarean section, most frequent indi- cations being CPD, previous caesarean section with failed trial of scar, fetal distress and failed induction of labour, who had no contraindications to general or spinal anaesthesia. Exclusion criteria: women who were unwilling, those not having a standardised technique of spinal or epidural, those with shock, bleeding disorders/haemorrhage, sepsis, severe hypertension and fixed cardiac output. Women with allergies to any of the anaesthetic drugs used, relative contraindications such as kyphosis or massive obesity and those with abnormal delays in theatre. Setting: South Africa.		
Interventions	Regional group had spinal anaesthesia with Heavy Marcaine. General anaesthesia group had Pentothal, scoline, mixture of nitrous oxide and oxygen, halothane and droperidol and fentanyl		
Outcomes	Study outcome was time taken to produce surgical readiness which was not one of the review outcomes thus study data not included in the review		
Notes			
Risk of bias			
Bias	Authors' judgement	Support for judgement	
Random sequence generation (selection bias)	Low risk	Randomly selected by picking numbers from a hat.	
Allocation concealment (selection bias)	High risk	No attempt was made to conceal allocation.	
Blinding (performance bias and detection bias) All outcomes	High risk	No blinding done.	
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.	
Selective reporting (reporting bias)	Unclear risk	Not stated.	

High risk

Regional versus general anaesthesia for caesarean section (Review)

Other bias

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Author states preference for spinal over gen-

eral anaesthesia.

Datta 1983

Methods	Randomisation was said to be by formal randomisation. Blinding of intervention and outcome: not stated. There was no loss to follow-up. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)	
Participants	30 women (10 received epidural, 10 received spinal and 10 received general anaesthesia) Inclusion criteria: healthy parturients, elective caesarean section. Exclusion criteria: none stated. Setting: USA, University Hospital.	
Interventions	2 types of regional anaesthesia were used in this study - spinal anaesthesia which had 0. 5% tetracaine. General anaesthesia group had thiopental with 50% nitrous oxide in oxygen	
Outcomes	Review measured neonatal umbilical arterial and venous pH, Apgar score less than 7 at 1 minute Study outcomes were I-D interval, UI-D interval, Apgar score < 7, maternal pH, neonatal acid-base values	
Notes		
Risk of bias		
Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Methods	Randomisation was said to be by a computer-generated table. Blinding of intervention and outcome: not stated. There was some loss to follow-up. gestational age group - 5 women excluded due to intraoperative transfusion, use of postoperative non-steroidal anti-inflammatory drugs and faulty sample withdrawal; Regional group - 5 women due to delay in blood sampling, epidural catheter dysfunction, haemolysis of blood samples and technical laboratory problems. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	25 women (17 received combined spinal-epidural, 18 received general anaesthesia). In- clusion criteria: uncomplicated elective caesarean section. Exclusions: multiple preg- nancy, medical problems in pregnancy, weight gain of more than 20 kg in pregnancy, predicted airway difficulty and contra-indication to a specific anaesthetic technique. Setting: Greece, University Hospital.
Interventions	Regional anaesthesia group had combined spinal and epidural anaesthesia with levobupi- vacaine and fentanyl +/- levobupivacaine respectively. General anaesthesia group had thiopentone, succinylcholine, nitrous oxide/oxygen and sevoflurane, and vecuronium
Outcomes	Study outcomes measured were median Apgar scores at 1 and 5 minutes but this review measures mean Apgar scores The study also measured IL-6 and TNF-alpha levels which are not part of the review outcomes
Notes	

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Computer generation.
Allocation concealment (selection bias)	Low risk	Sealed envelopes.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

992

Methods	Women said to have been randomised but method not stated. Blinding of intervention: not stated. Blinding of outcome assessment: the paediatrician was blind to the type of anaesthesia used. No loss to follow-up. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	47 women (23 received epidural, 24 received general anaesthesia). Inclusion criteria: elective caesarean section for breech presentation or disproportion following a normal uncomplicated pregnancy. Exclusions: none stated. Setting: Germany, University Hospital.
Interventions	Regional anaesthesia group had epidural anaesthesia with 12-15 mL of bupivacaine. General anaesthesia group had thiopentone, succinylcholine, nitrous oxide/oxygen and halothane
Outcomes	Outcomes measured: study measured maternal heart rate, blood pressure, blood gases and haematocrit. Induction-delivery and I-D interval were also measured. Neonatal Apgar scores, umbilical arterial and venous blood gases and acid-base balance and a full set of neurological observations were also measured. Review measured neonatal umbilical arterial and venous pH and Apgar scores of 4 or less at 1 and 5 minutes
Notes	The outcomes measured in the review were those that were reported clearly in the results

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Dogan 2008

Methods	Women said to have been randomised but method not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. No loss to follow-up. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	40 women (20 received combined spinal-epidural, 20 general anaesthesia). Inclusion criteria: elective caesarean section in term women in ASA groups I and II. Exclusions: none stated. Setting: Baskent University Hospital, Turkey.
Interventions	Regional anaesthesia group had combined spinal-epidural anaesthesia with bupivacaine. General anaesthesia group had thiopental sodium, atracurium, nitrous oxide/oxygen and sevoflurane
Outcomes	Outcomes measured: study measured outcomes that were different from the outcomes of this review. These included maternal heart rate, blood pressure, ST segment changes on electrocardiogram, serum troponin T, CK-MB and myoglobin concentrations
Notes	

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Methods	Women were said to have been randomised by sealed envelopes. Blinding of intervention: not stated. Blinding of outcome assessment: the paediatrician was blinded to the type of anaesthesia used. No mothers were excluded but there were no data for 1 neonate in the general anaesthesia group as its mother suffered a stillbirth. Intention to treat: not stated (but all women remained in their allocated groups)
Participants	70 women (35 received spinal, 35 received general anaesthesia). Inclusion criteria: pre-eclampsia with non-reassuring fetal heart trace. Exclusion criteria: participant refusal; any other relative contraindication to general or spinal anaesthesia, in particular oral intake other than clear fluids within 4 hours of the intended surgery; body mass index greater than 35 kg/m2; Mallampati score greater than 2; clinical signs of hypovolaemia; abruptio placentae; placenta praevia; coagulation abnormality; thrombocytopenia; local or generalised sepsis; spinal deformity; cord pro- lapse; less than 30 weeks' gestation; or twin pregnancy. Setting: South Africa.
Interventions	Regional group had spinal anaesthesia using 1.8 mL hyperbaric bupivacaine 0.5% with 10 ug fentanyl. General anaesthesia group had thiopentone, suxamethonium, nitrous oxide/oxygen, isoflurane and magnesium sulphate to control the pressor response to tracheal intubation
Outcomes	Outcomes measured: study primary outcomes were umbilical arterial base deficit, um- bilical arterial pH, Apgar scores, requirements for resuscitation, and complications. Sec- ondary outcome measures were maternal pulse rate and non-invasive blood pressure. Review measured umbilical arterial pH, need for oxygen by face mask or intubation, and maternal blood loss

### Notes

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Low risk	Sealed envelopes.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.

# Dyer 2003 (Continued)

Other bias	Unclear risk	Not stated.
Hodgkinson 1980		
Methods	Women were said to be randomly allocated Blinding of intervention: not stated. Blinding of outcome assessment: not stated No women were excluded. Intention-to-treat analysis: not stated (but al	to groups but the method was not stated. Il women remained in their allocated groups)
Participants	20 women (10 received epidural, 10 receive Inclusion criteria: severe gestational hypert superimposed pre-eclampsia) requiring eme Exclusion criteria: not stated. Setting: USA, University Health Science Co	ed general anaesthesia). ension (pre-eclampsia or hypertension with ergency caesarean section for delivery. entre.
Interventions	Regional group had epidural anaesthesia wit anaesthesia group had thiopentone, succiny	h 12-20 mL of bupivacaine 0.75%. General l choline, nitrous oxide and halothane
Outcomes	Outcomes measured: study primary outcom sures before, during and after surgery. Rev neonatal Apgar scores at 1 and 5 minutes	es were systemic and pulmonary blood pres- iew measured maternal adverse events and
Notes		

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Methods	Inadequate randomisation as women were allocated to groups alternately. Blinding of intervention: not stated. Blinding of assessment: the examiner who performed the neonatal neurologic assessment was blinded to the obstetric and anaesthetic management. No women were excluded. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	30 women (15 received epidural, 15 received general anaesthesia). Inclusion criteria: healthy women with uncomplicated full-term pregnancies, requiring elective caesarean section. 3 women in each group had mild toxemia, diabetes or hyper- tension and 1 person in the general anaesthesia group had partial placenta previa and transverse lie. All the women had intact membranes and were not in labour. Exclusion criteria: not stated. Setting: Finland; type of hospital not stated.
Interventions	Regional group had epidural anaesthesia with lidocaine and epinephrine. General anaes- thesia group had thiopentone, 1:1 mixture of nitrous oxide and oxygen, and succinyl choline
Outcomes	Outcomes measured: study outcome measures were maternal and fetal blood gases, neonatal Apgar scores and neurological assessment. Review measured neonatal umbilical vessel pH and neurological assessment
Notes	

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	High risk	Allocation sequence quasi-randomised.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Low risk	The person who performed the neonatal neurologic assessment was blinded to the obstetric and anaesthetic management
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Hong 2002

Methods	Study was said to be a randomised trial but the method was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: an anaesthesiologist blinded to the anaesthetic tech- nique measured the estimated blood loss, volume of IV fluids and blood given, and the Apgar scores of the newborn. No women were excluded. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	25 women (13 received epidural, 12 received general anaesthesia). Inclusion criteria: women with grade 4 placenta previa without bleeding, scheduled for elective caesarean section. Exclusion criteria: not stated. Setting: South Korea, University Hospital.
Interventions	Regional anaesthesia group had epidural anaesthesia with lidocaine (20 mL of 2%), plus epinephrine (1 in 200,000) and morphine (2 mg in 4 mL). General anaesthesia group had thiopentone, succinyl choline, vecuronium, mixture of nitrous oxide and oxygen and enflurane
Outcomes	Outcomes measured: study outcome measures were maternal blood pressure and heart rate, estimated blood loss at surgery, and neonatal Apgar scores, haemoglobin and haema- tocrit levels at admission and 24 hours after surgery, need for postoperative transfusion, request for analgesics and adverse events. Review measured amount of blood received, need for postoperative blood transfusion, maternal estimated blood loss, need for postoperative analgesia, time to request analgesia, adverse events and Apgar scores

Notes

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Jain 2009

Methods	Study was said to be a randomised trial but the method was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. No women were excluded. Intention-to-treat analysis: not stated.
Participants	40 women (20 received spinal, 20 received general anaesthesia). Inclusion criteria: women with proven uteroplacental insufficiency. Exclusion criteria: not stated. Setting: India.
Interventions	Regional anaesthesia group had spinal anaesthesia with heavy bupivacaine with 25 ug Fentanyl. General anaesthesia group had thiopentone and suxamethonium.
Outcomes	Outcomes measured: study outcome measures were maternal blood pressure, median Apgar scores at 1 and 5 minutes, mean umbilical artery pH and mean umbilical vein pH, neonatal need for oxygen and resuscitation
Notes	Study was reported in an abstract; numbers in each group were not stated so comparisons could not be made. The data were therefore not included in the review. Study authors were emailed with no response

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Methods	Women were said to have been allocated randomly to groups but the method of ran- domisation was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. 20 women were excluded; 19 due to incomplete data on their infants and 1 due to congenital malformation in her infant. Intention-to-treat analysis: not stated but women remained in their allocated groups
Participants	104 women (46 received spinal, 38 received general anaesthesia and 20 were excluded; 19 due to incomplete data and 1 for minor fetal congenital malformation). Inclusion criteria: healthy women with uncomplicated singleton cephalic pregnancies undergoing elective repeat caesarean section after 37 weeks' gestation. Exclusion criteria: pregnancies with obstetric or medical complications. Setting: Turkey, hospital not stated.
Interventions	Regional anaesthesia group had spinal anaesthesia with 12.5 mg of 0.5% heavy bupiva- caine and morphine. General anaesthesia group had thiopental sodium, succinyl choline, mixture of nitrous oxide and oxygen, sevoflurane and vecuronium
Outcomes	Outcomes measured: study primary endpoints were respiratory depression, perinatal asphyxia, readmission and duration of hospital admission of the infants. Review measured umbilical arterial pH, need for neonatal oxygen therapy and Apgar scores
Notes	
Risk of bias	

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Methods	Study was said to be a randomised trial but the method was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. No women were excluded. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)		
Participants	30 women (20 received epidural,10 received general anaesthesia). Inclusion criteria: healthy women undergoing caesarean section. Exclusion criteria: not stated. Setting: South Korea, University College of Medicine.		
Interventions	Regional anaesthesia group had epidural anaesthesia with bupivacaine. General anaesthesia group had sodium thiopental, succinyl choline, enflurane, mixture of nitrous oxide and oxygen, midazolam and vencuronium		
Outcomes	Outcomes measured: study outcome measures were changes in serum interleukin and cortisol levels		
Notes	Study outcomes were not relevant to the review so the data were not included		
Risk of bias			
Bias	Authors' judgement	Support for judgement	
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.	
Allocation concealment (selection bias)	Unclear risk	Not stated.	
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.	
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.	
Selective reporting (reporting bias)	Unclear risk	Not stated.	
Other bias	Unclear risk	Not stated.	

# Korkmaz 2004

Methods

Study was said to be a randomised trial but the method was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. No other method was stated. This was an abstract

Regional versus general anaesthesia for caesarean section (Review)

# Korkmaz 2004 (Continued)

Participants	30 women (15 received combined regional, 15 received general anaesthesia). Inclusion criteria: women who had elective caesarean section. Exclusion criteria: not stated. Setting: Turkey, Education and Research Hospital.
Interventions	Regional group had combined spinal and epidural anaesthesia with 5 mg of 0.5% bupi- vacaine and fentanyl, with additional ropivacaine top-ups if necessary. The general anaes- thesia group had sevoflurane
Outcomes	Study outcome measures were heart rate, mean blood pressure, systolic and diastolic blood pressure, Apgar scores at 1 and 5 minutes and umbilical blood gases Review measured Apgar scores at 1 and 5 minutes. Blood gas levels were not reported in the abstract
Notes	

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

# Lertakyamanee 1999

Methods	Randomisation was said to be by random numbers.
	Blinding of intervention: not stated.
	Blinding of outcome assessment: the assessors of the estimated blood loss levels, haema-
	tocrit levels and NACS were blinded to the mode of anaesthesia used.
	There were no women excluded.
	Intention-to-treat analysis: not stated but 39 women were changed to another interven-
	tion group due to technical difficulty

# Lertakyamanee 1999 (Continued)

Participants	<ul> <li>341 women (120 received epidural, 118 received spinal and 103 received general anaesthesia)</li> <li>Inclusion criteria: term normal women scheduled to have elective or emergency caesarean section.</li> <li>Exclusion criteria: women with abruptio placenta, bleeding placenta praevia, fetal distress, diabetes mellitus, moderate to severe hypertension of pregnancy, severe cardiac or respiratory disease, pregnancy with more than 1 fetus and coagulopathy.</li> <li>Setting: Department of Obstetrics and Gynaecology, Thailand, University Hospital</li> </ul>
Interventions	2 types of regional anaesthesia were used in this study - spinal anaesthesia group which used 5% lidocaine and epidural which used 2% lidocaine. The general anaesthesia group had halothane, a mixture of nitrous oxide and oxygen and pancuronium bromide
Outcomes	Study outcome measures were estimated blood loss, IV fluid and blood transfusion, pre and postoperative haematocrit, intraoperative complications, hypo and hypertension, satisfaction towards anaesthetic technique and total pain scores. Apgar scores, umbilical vein gases, neurologic adaptive capacity scores, and maternal systolic blood pressure Review measured difference between pre and postoperative haematocrit, number who had blood transfusion, hypo and hypertension, maternal satisfaction with technique, intraoperative pain, nausea and vomiting, maternal systolic blood pressure, umbilical venous pH, neonatal neurologic and adaptive capacity scores and Apgar scores at 1 and 5 minutes

Notes

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Allocation sequence by random numbers.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Low risk	Those estimating blood loss and haemat- ocrit levels and assessors of the NACS were blinded to the type of anaesthesia used
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Methods	Randomisation was adequate (done with a random chart). Blinding of intervention: not stated. Blinding of outcome assessment: the paediatrician who assessed the neonatal NACS was blinded to the anaesthetic technique used. No women were excluded. Intention-to-treat analysis: not stated but all women remained in their allocated groups
Participants	90 women (30 received epidural, 30 received spinal and 30 received general anaesthesia) Inclusion criteria: healthy women presenting for elective caesarean section, at a gestational age greater than 36 weeks, with infants of a birthweight greater than 2.5 kg, with no evidence of placental insufficiency. Exclusion criteria: not stated. Setting: India; type of hospital not stated.
Interventions	2 types of regional anaesthesia were used in this study - epidural anaesthesia which used 0.5% bupivacaine, and spinal with 1% bupivacaine. General anaesthesia group had thiopentone, suxamethonium, nitrous oxide and oxygen, halothane and pancuronium
Outcomes	Study outcome measures were maternal blood pressure and heart rate, maternal blood gases, umbilical arterial and venous blood gases, time intervals to delivery, Apgar scores and neonatal NACS. Review measured umbilical arterial and venous pH, neonatal NACS and Apgar scores
Notes	

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Allocation sequence by random numbers chart.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Low risk	The assessors of the NACS were blinded to the mode of anaesthesia used
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Mancuso 2	01	0
-----------	----	---

Methods	Study was appropriately randomised by computer-generated random numbers. Blinding of intervention: not stated. Blinding of outcome assessment: a paediatrician blinded to the anaesthetic techniques estimated the umbilical artery pHs and the Apgar scores of the newborn. 8 infants were excluded but attrition was still less than 10%. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	179 women (90 received spinal, 89 received general anaesthesia). Inclusion criteria: women with previous caesarean section, maternal-fetal disproportion, restricted pelvis, patient's choice, breech presentation, history of sterility and maternal age, who were scheduled for elective caesarean section. Exclusion criteria: those without the above indications. Setting: Messina, Italy; University Hospital.
Interventions	Regional anaesthesia group had spinal anaesthesia with hyperbaric bupivacaine or isobaric levobupivacaine. General anaesthesia group had propofol, cis-atracurium, sevoflurane and mixture of nitrous oxide and oxygen
Outcomes	Outcomes measured: study outcome measures were neonatal Apgar scores, umbilical artery pH and need for oxygen by face mask
Notes	

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Allocation sequence generated by computer.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Low risk	The assessors of the umbilical artery pH and Apgar scores were blinded to the mode of anaesthesia used
Incomplete outcome data (attrition bias) All outcomes	Low risk	Not stated.
Selective reporting (reporting bias)	Low risk	Not stated.
Other bias	Unclear risk	Not stated.

# Mathur 2002

Methods	Study was said to be a randomised trial but the method was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. None of the women were excluded. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	60 women (20 received epidural, 20 received spinal and 20 received general anaesthesia) Inclusion criteria: women with pre-eclampsia undergoing caesarean section for various indications other than fetal distress. Exclusion criteria: women with medical complica- tions, eclampsia and platelets > 100,000/mm <sup>3</sup> . Setting: India, S.N. Medical College, Agra.
Interventions	2 types of regional anaesthesia were used in this study - spinal anaesthesia group which used bupivacaine and epidural which used 2% lidocaine or bupivacaine. The general anaesthesia group had pentothal and succinyl choline, and were maintained on nitrous oxide and oxygen
Outcomes	Study outcome measures were anaesthesia induction to skin incision, skin incision to delivery, uterine incision to delivery, surgery duration, highest and lowest maternal blood pressure, IV fluid input and urinary output, perioperative blood loss, maternal complications such as hyper- and hypo- tension, vomiting, aspiration, postpartum haemorrhage and post spinal headache. Pulmonary oedema, cerebral haemorrhage, convulsions and maternal mortality were also outcome measures but there were no reported incidences Review measured mean Apgar scores at 1 and 10 minutes, and vomiting as an adverse event in the mothers

# Notes

# Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	High risk	There were errors in the paper such as the level of platelet count excluded (> 100,000
instead of < 100,000) as well as some other errors in the tables. There were also a number of typographical errors in the paper Momani 2001 Methods Study was said to be a randomised trial but the method was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. None of the women were excluded. Intention-to-treat analysis: not stated (but all women remained in their allocated groups) Participants 45 women (15 received spinal, 15 received general with bupivacaine supplementation and 15 received general anaesthesia without bupivacaine). Inclusion criteria: healthy women and those with mild systemic disease undergoing elective or emergency caesarean section. Setting: Prince Hashim Ben Al-Hussein Hospital, Amman, Jordan There were 2 groups of general anaesthesia in this study - general anaesthesia with bupi-Interventions vacaine wound infiltration and general anaesthesia without. Thiopentone, suxamethonium, halothane and 50% nitrous oxide in oxygen were used for the general anaesthesia. The regional anaesthesia was spinal anaesthesia group which used hyperbaric bupivacaine; bupivacaine was also infiltrated into the wound Study outcome measures were reported in a different manner from the review outcome Outcomes measures and could not be extrapolated or re-calculated thus the data could not be used. They included time to next analgesia (recorded as a range as opposed to mean or median) and pain degree on a visual analogue scale recorded in 7 different time slots including zero and 24 hours after the operation

Notes

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.

Regional versus general anaesthesia for caesarean section (Review)

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

#### Momani 2001 (Continued)

Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

#### Moslemi 2007

Methods	Study was said to be a randomised trial but the method was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. None of the women were excluded. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	60 women, 62 neonates (including 2 sets of twins) (30 received spinal, 30 received general anaesthesia). Inclusion criteria: women with severe pre-eclampsia undergoing caesarean section. Setting: Iran, Tabriz University.
Interventions	Regional anaesthesia group had spinal anaesthesia with hyperbaric bupivacaine and fen- tanyl. General anaesthesia group had thiopental, succinyl choline, lidocaine, fentanyl, mixture of nitrous oxide and oxygen, halothane and atracurium
Outcomes	Study outcome measures were maternal nausea and vomiting, maternal blood pressure changes, neonatal umbilical artery pH and neonatal Apgar scores
Notes	

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Nabhan 2009

Methods	Study was appropriately randomised by computer-generated software. Blinding of intervention: not blinded. Blinding of outcome assessment: blinded. None of the women were excluded. Intention-to-treat analysis: analysis was by intention to treat
Participants	82 women (41 received spinal, 41 received general anaesthesia). Inclusion criteria: healthy women for elective caesarean section with singleton normal pregnancies Exclusion criteria: women with intrauterine infection, chromosomal aberration, major malformations, non-reassuring fetal heart rate pattern, antepartum haemorrhage, dia- betic and pre-eclamptic pregnancy and cardiac disease. Setting: Egypt; Ain Shams University, Cairo.
Interventions	Regional anaesthesia group had spinal anaesthesia with hyperbaric bupivacaine and fen- tanyl. General anaesthesia group had sodium thiopental, succinyl choline, isoflurane and mix- ture of nitrous oxide and oxygen
Outcomes	Study outcome measures were umbilical venous glutathione concentration, umbilical venous malondialdehyde, umbilical venous pH, pO2, PCO2, neonatal Apgar scores, neonatal morbidity and admission to neonatal intensive care unit
Notes	

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Allocation sequence generated by computer.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

#### Papadopoulou 2005

Methods	Study was said to be a randomised trial but the method used was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. None of the women were excluded. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	32 women (14 received spinal, 18 received general anaesthesia). Inclusion criteria: women requiring emergency caesarean section. Setting: Hippokrates General Hospital, Thessaloniki, Greece.
Interventions	Regional group had spinal anaesthesia with 10-15 mg bupivacaine. General anaesthesia group had thiopentone and 'Sucin' with sevoflurane and nitrous oxide 50% in oxygen
Outcomes	Study outcome measures that were documented (b-endorphin and cortisol concentra- tion) were different from those measured in the review. Other study outcomes were maternal heart rate and arterial blood pressure, acid base balance of both mothers and newborns and newborn Apgar scores

Notes

### Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

#### Pence 2002

Methods

Study was said to be a randomised trial but the method was not stated.Blinding of intervention: not stated.Blinding of outcome assessment: not stated. No women were excluded.Intention-to-treat analysis: not stated (but all women remained in their allocated groups)

Regional versus general anaesthesia for caesarean section (Review)

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

#### Pence 2002 (Continued)

Participants	56 women (26 received epidural, 30 receive Inclusion criteria: women with a cephalic pro Exclusion criteria: women with medical dise iron. Setting: Turkey.	d general anaesthesia). esentation, having elective caesarean section. ases, fetal distress and medication apart from
Interventions	Regional group had epidural anaesthesia with 50 mg bupivacaine and fentanyl. General anaesthesia group had isoflurane with propofol and succinylcholine	
Outcomes	Study outcome measures were umbilical artery blood gas levels and malondialdehyde and glutathione levels Review measured umbilical arterial pH.	
Notes		
Risk of bias		
Bias	Authors' judgement	Support for judgement

Dias	Ruthors judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

#### Petropoulos 2003

Methods	Randomisation was done with a random-number table and numbered sealed envelopes were used to conceal allocation. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. There was no loss to follow-up. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	230 women (72 received epidural, 78 received spinal and 80 received general anaesthesia) Inclusion criteria: pregnant women presenting for planned elective caesarean section after 38 weeks' gestation.

#### **Petropoulos 2003** (Continued)

	Exclusion criteria: multiple gestation, gestational age < 38 weeks and > 42 weeks, placental or cord abnormalities, premature rupture of membranes, abnormal fetal heart tracings, obstetric or medical complications, congenital malformations and incomplete data. Setting: Greece, University Hospital.
Interventions	2 types of regional anaesthesia were used in this study - epidural anaesthesia which used ropivacaine after a test dose of xylocaine. General anaesthesia group had thiopentone, suxamethonium,nitrous oxide and oxygen, sevoflurane and vecuronium
Outcomes	Outcomes measured: study outcomes were maternal blood gases, neonatal blood gases, Apgar scores and need for oxygen or mask ventilation. Review measured neonatal umbilical artery pH, Apgar scores at 1 and 5 minutes less than 7, and need for oxygen or mask ventilation of the neonate

Notes

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Allocation sequence computer-generated.
Allocation concealment (selection bias)	Low risk	Sealed envelopes.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

### Turhanoglu 1999

Methods	Study was said to be a randomised trial but the method was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. No women were excluded. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	60 women (30 received spinal, 30 received general anaesthesia). Inclusion criteria: pregnant women with hypertension requiring a caesarean section for delivery. Exclusion criteria: women with hypertension existing before pregnancy, thrombocyte

#### Turhanoglu 1999 (Continued)

	count less than 100,000 and with 'extremely damaged fetal parameters'
Interventions	The regional anaesthesia group had spinal anaesthesia with 12.5 mg of 0.5% heavy bupi- vacaine. General anaesthesia group had propofol, succinyl choline, mixture of nitrous oxide and oxygen and isoflurane. Vecuronium was also added where necessary
Outcomes	Study outcomes were maternal heart rate and arterial blood pressures before during and after surgery, uterine incision-cord clamping interval, postoperative pain scores, time to first postoperative analgesia, and 1 and 5 minute Apgar scores Review measured neonatal umbilical artery pH, Apgar scores at 1 and 5 minutes less than 7, and need for oxygen or mask ventilation of the neonate

#### Notes

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

#### Wallace 1995

Methods	Randomisation was by a random-number table and numbered sealed envelopes were used to conceal allocation. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. 1 woman was excluded from the study after randomisation. Intention-to-treat analysis: not stated. All the remaining women stayed in their allocated groups
Participants	80 women (27 received epidural, 27 received combined spinal-epidural and 26 received general anaesthesia). Inclusion criteria: women undergoing elective or emergency caesarean section for severe pre-eclampsia.

#### Wallace 1995 (Continued)

	Exclusion criteria were thrombocytopenia with a platelet count of less than 100,000/mm <sup>3</sup> , eclampsia or medical conditions such as heart disease, diabetes mellitus or chronic renal disease, and non-reassuring fetal heart trace. Setting: Labor and Delivery Unit, USA, University Hospital.
Interventions	2 types of regional anaesthesia were used in this study - epidural anaesthesia which used 2% lidocaine or 3% chloroprocaine. General anaesthesia group had pentothal, succinylcholine, mixture of nitrous oxide and oxygen, isoflurane and atracurium or vecuronium. Lidocaine and nitroglycerin were also administered before intubation to prevent hypertension from tracheal stimulation
Outcomes	Outcomes measured: study outcomes were maternal systolic and diastolic blood pres- sures, time intervals of preparation for anaesthesia, and surgical and delivery events, IV fluid volumes administered and urine output, neonatal gestational age, birthweight, Apgar scores, umbilical artery blood gases, admission to special care nursery, incidence of small-for-gestational-age infants, those with respiratory distress requiring mechanical ventilation and those with intracranial haemorrhage. Review measured highest and lowest intraoperative blood pressures, umbilical artery pH and Apgar scores

#### Notes

### Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence computer-generated.
Allocation concealment (selection bias)	Low risk	Sealed envelopes.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Yegin 2003

Methods	Study was said to be a randomised trial but the method was not stated. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. No women were excluded. Intention-to-treat analysis: not stated (but all women remained in their allocated groups)
Participants	62 women (31 received epidural, 31 received general anaesthesia). Inclusion criteria: uncomplicated women who were to give birth at term and classified as ASA I or II. Exclusion criteria: not stated. Setting: Turkey; hospital not stated.
Interventions	Regional group had epidural anaesthesia with 15 mL of 0.5% bupivacaine. The general group had isoflurane with vecuronium, thiopental and suxamethonium
Outcomes	Outcomes measured: study measured umbilical arterial and venous blood gases and mean Apgar scores at 1 and 5 minutes. Review measured umbilical arterial and venous pH and mean Apgar scores at 1 and 5 minutes
Notes	

Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

Yentur 2009

Methods	Study was appropriately randomised by computer-generated software. Blinding of intervention: not stated. Blinding of outcome assessment: not stated. 7 of the 70 women were excluded. Intention-to-treat analysis: not stated but women remained in their allocated groups
Participants	70 women (35 received epidural, 35 received general anaesthesia). Inclusion criteria: low-risk pregnant women, 18 years or older, 37 weeks' gestation or more that were scheduled for elective caesarean section Exclusion criteria: women with pre-eclampsia or eclampsia, heart disease, taking chronic medication or other high-risk pregnancies. Setting: Turkey; Celal Bayal University, Manisa.
Interventions	Regional anaesthesia group had epidural anaesthesia with bupivacaine. General anaesthesia group had thiopental, succinyl choline, isoflurane and mixture of nitrous oxide and oxygen and atracurium
Outcomes	Outcomes measured: study measured maternal rectal temperature, neonatal rectal tem- perature, umbilical vein pH and neonatal Apgar scores
Notes	
Risk of bias	

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Allocation sequence not stated.
Allocation concealment (selection bias)	Unclear risk	Not stated.
Blinding (performance bias and detection bias) All outcomes	Unclear risk	Not stated.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Not stated.
Selective reporting (reporting bias)	Unclear risk	Not stated.
Other bias	Unclear risk	Not stated.

ASA: American Society of Anaesthesiologists classification CK: creatine kinase CPD: cephalopelvic disproportion I-D: incision-delivery IV: intravenous kg/m2: kilogram per metre squared

### Characteristics of excluded studies [ordered by study ID]

Study	Reason for exclusion
Abboud 1985	Randomisation was not done; the women were divided into 3 unequal groups
Akturk 1995	Randomisation was not done; women were assigned to either group according to their individual preference
Fyneface-Ogan 2008	There was no regional anaesthesia group. Instead local anaesthesia was compared with general anaesthesia for caesarean section
Gambling 1995	Randomisation was confined only to the general anaesthesia groups; the women in the spinal anaesthesia group were assigned on request
Kamat 1991	No mention of randomisation.
Navarro 2000	Randomisation was confined to the general anaesthesia groups; the women in the spinal anaesthesia group were assigned on request
Qublan 2001	No randomisation done; the type of anaesthesia was chosen by the woman in consultation with the anaes- thesiologist
Ratcliffe 1992	No randomisation done; women were allowed to choose between regional and general anaesthesia
White 1962	Participating obstetricians requested that all the women have spinal anaesthesia towards the end of the study, resulting in an unequal distribution of cases

#### DATA AND ANALYSES

Outcome or subgroup title	No. of No. of studies participants		Statistical method	Effect size	
1 Mean umbilical arterial pH	8	454	Mean Difference (IV, Random, 95% CI)	-0.00 [-0.02, 0.01]	
1.1 Non-urgent indication for caesarean section	7	397	Mean Difference (IV, Random, 95% CI)	0.00 [-0.01, 0.02]	
1.2 Urgent indication for caesarean section: severe pre-eclampsia	1	57	Mean Difference (IV, Random, 95% CI)	-0.04 [-0.07, -0.01]	
2 Mean umbilical venous pH	7	505	Mean Difference (IV, Random, 95% CI)	0.01 [-0.01, 0.02]	
3 Apgar score of 4 or less at 1 minute	1	47	Risk Ratio (M-H, Fixed, 95% CI)	0.15 [0.01, 2.73]	
4 Apgar score of 4 or less at 5 minutes	1	47	Risk Ratio (M-H, Fixed, 95% CI)	0.35 [0.01, 8.11]	
5 Apgar score of 6 or less at 1 minute (not prespecified in protocol)	2	209	Risk Ratio (M-H, Fixed, 95% CI)	0.58 [0.22, 1.52]	
5.1 Non-urgent indication for caesarean section	1	152	Risk Ratio (M-H, Fixed, 95% CI)	0.67 [0.17, 2.69]	
5.2 Urgent indication for caesarean section: severe pre-eclampsia	1	57	Risk Ratio (M-H, Fixed, 95% CI)	0.50 [0.13, 1.91]	
6 Apgar score of 6 or less at 5 minutes (not prespecified in protocol)	2	209	Risk Ratio (M-H, Fixed, 95% CI)	0.46 [0.11, 1.95]	
6.1 Non-urgent indication for caesarean section	1	152	Risk Ratio (M-H, Fixed, 95% CI)	0.74 [0.13, 4.31]	
6.2 Urgent indication for caesarean section: severe pre-eclampsia	1	57	Risk Ratio (M-H, Fixed, 95% CI)	0.17 [0.01, 3.37]	
7 Mean Apgar score at 1 minute	5	408	Mean Difference (IV, Random, 95% CI)	0.39 [-0.34, 1.12]	
8 Mean Apgar score at 5 minutes	4	368	Mean Difference (IV, Random, 95% CI)	0.20 [-0.15, 0.55]	
9 Neonatal neurologic and adaptive capacity score at 2-4 hours	2	253	Mean Difference (IV, Random, 95% CI)	2.17 [-1.13, 5.47]	
10 Amount of blood transfusion received in units (not prespecified in protocol)	1	25	Mean Difference (IV, Fixed, 95% CI)	-0.70 [-1.73, 0.33]	
11 Number who received postoperative blood transfusion (not prespecified in protocol)	2	256	Risk Ratio (M-H, Random, 95% CI)	0.80 [0.23, 2.76]	
12 Maternal estimated blood loss in mL	2	256	Std. Mean Difference (IV, Fixed, 95% CI)	-0.32 [-0.56, -0.07]	
13 Difference between pre and postoperative haematocrit (%)	1	231	Mean Difference (IV, Fixed, 95% CI)	1.70 [0.47, 2.93]	

#### Comparison 1. Epidural versus general anaesthesia

14 Satisfaction score on visual analogue scale	1	223	Mean Difference (IV, Fixed, 95% CI)	-0.01 [-0.63, 0.61]
15 Number who would prefer the same technique again	1	223	Risk Ratio (M-H, Fixed, 95% CI)	0.80 [0.65, 0.98]
16 Neonatal neurologic and adaptive capacity score of < 35 at 15 minutes	1	60	Risk Ratio (M-H, Fixed, 95% CI)	0.94 [0.62, 1.45]
17 Neonatal neurologic and adaptive capacity score of < 35 at 2 hours	1	60	Risk Ratio (M-H, Fixed, 95% CI)	0.67 [0.27, 1.64]
18 Adverse events	5		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
18.1 Headache	1	20	Risk Ratio (M-H, Random, 95% CI)	1.2 [0.54, 2.67]
18.2 Epigastric pain	1	20	Risk Ratio (M-H, Random, 95% CI)	0.14 [0.01, 2.45]
18.3 Blurred vision	1	20	Risk Ratio (M-H, Random, 95% CI)	2.0 [0.47, 8.56]
18.4 Convulsion	1	20	Risk Ratio (M-H, Random, 95% CI)	1.0 [0.17, 5.77]
18.5 Nausea	3	286	Risk Ratio (M-H, Random, 95% CI)	1.82 [0.26, 12.80]
18.6 Vomiting	3	301	Risk Ratio (M-H, Random, 95% CI)	0.86 [0.14, 5.25]
18.7 Pruritus	1	25	Risk Ratio (M-H, Random, 95% CI)	8.36 [0.50, 140.56]
18.8 Shivering	1	30	Risk Ratio (M-H, Random, 95% CI)	0.13 [0.02, 0.88]
18.9 Allergic reaction	1	30	Risk Ratio (M-H, Random, 95% CI)	0.11 [0.01, 1.90]
18.10 Bradycardia	1	30	Risk Ratio (M-H, Random, 95% CI)	1.5 [0.29, 7.73]
19 Need for oxygen therapy or mask ventilation of the neonate	1	152	Risk Ratio (M-H, Fixed, 95% CI)	0.86 [0.34, 2.20]
20 Maternal blood loss > 500 mL	1	40	Risk Ratio (M-H, Fixed, 95% CI)	0.5 [0.10, 2.43]
21 Mean Apgar score at 10 minutes	1	40	Mean Difference (IV, Fixed, 95% CI)	-0.45 [-1.03, 0.13]
22 Time to request postoperative analgesia in minutes	1	25	Mean Difference (IV, Fixed, 95% CI)	500.0 [364.36, 635. 64]
23 Intraoperative pain score on visual analogue scale	1	223	Mean Difference (IV, Fixed, 95% CI)	0.84 [0.45, 1.23]

### Comparison 2. Spinal versus general anaesthesia

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Umbilical arterial pH	6	459	Mean Difference (IV, Fixed, 95% CI)	-0.00 [-0.01, 0.00]
2 Umbilical venous pH	4	383	Mean Difference (IV, Fixed, 95% CI)	0.01 [-0.00, 0.02]
3 Neonatal neurologic and adaptive capacity score at 2-4 hours	1	221	Mean Difference (IV, Fixed, 95% CI)	0.40 [-0.54, 1.34]
4 Neonatal neurologic and adaptive capacity score of < 35 at 15 minutes	1	60	Risk Ratio (M-H, Fixed, 95% CI)	0.17 [0.05, 0.51]
5 Neonatal neurologic and adaptive capacity score of < 35 at 2 hours	1	60	Risk Ratio (M-H, Fixed, 95% CI)	0.05 [0.00, 0.87]

6 Neonatal neurologic and adaptive capacity score of < 35 at 24 hours	1	60	Risk Ratio (M-H, Fixed, 95% CI)	0.0 [0.0, 0.0]
7 Mean Apgar score at 1 minute	5	470	Mean Difference (IV, Random, 95% CI)	0.54 [-0.09, 1.16]
8 Mean Apgar score at 5 minutes	4	429	Mean Difference (IV, Random, 95% CI)	0.24 [-0.05, 0.54]
9 Apgar score of 6 or less at 1 minute (not prespecified in protocol)	5	435	Risk Ratio (M-H, Random, 95% CI)	0.54 [0.16, 1.84]
9.1 Non-urgent indication for caesarean section	4	373	Risk Ratio (M-H, Random, 95% CI)	0.50 [0.07, 3.39]
9.2 Urgent indication for caesarean section: severe pre-eclampsia	1	62	Risk Ratio (M-H, Random, 95% CI)	0.78 [0.27, 2.29]
10 Apgar score of 6 or less at 5 minutes (not prespecified in protocol)	3	291	Risk Ratio (M-H, Fixed, 95% CI)	0.56 [0.21, 1.46]
11 Maternal estimated blood loss in mL	2	279	Std. Mean Difference (IV, Fixed, 95% CI)	-0.59 [-0.83, -0.35]
12 Difference between pre and postoperative haematocrit	1	209	Mean Difference (IV, Fixed, 95% CI)	3.10 [1.73, 4.47]
13 Number who received postoperative blood transfusion (not prespecified in protocol)	1	209	Risk Ratio (M-H, Fixed, 95% CI)	0.3 [0.07, 1.38]
14 Number who would prefer the same technique again	1	221	Risk Ratio (M-H, Fixed, 95% CI)	0.80 [0.65, 0.99]
15 Satisfaction score on visual analogue scale	1	221	Mean Difference (IV, Fixed, 95% CI)	-0.58 [-1.26, 0.10]
16 Adverse events	3		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
16.1 Nausea	2	269	Risk Ratio (M-H, Random, 95% CI)	3.14 [0.21, 48.07]
16.2 Vomiting	3	310	Risk Ratio (M-H, Random, 95% CI)	1.88 [0.36, 9.69]
17 Maternal blood loss > 500 mL	1	41	Risk Ratio (M-H, Fixed, 95% CI)	0.24 [0.03, 1.95]
18 Mean Apgar score at 10 minutes	1	41	Mean Difference (IV, Fixed, 95% CI)	-0.15 [-0.81, 0.51]
19 Time to request postoperative analgesia in minutes	1	60	Mean Difference (IV, Fixed, 95% CI)	97.8 [90.28, 105.32]
20 Neonatal need for oxygen by mask or intubation	1	171	Risk Ratio (M-H, Fixed, 95% CI)	0.04 [0.00, 0.60]
21 Intraoperative pain score on visual analogue scale	1	221	Mean Difference (IV, Fixed, 95% CI)	0.69 [0.32, 1.06]

### Comparison 3. Combined spinal-epidural versus general anaesthesia

Outcome or subgroup title	No. of No. of come or subgroup title studies participants		Statistical method	Effect size
1 Umbilical arterial pH	2	211	Mean Difference (IV, Fixed, 95% CI)	-0.03 [-0.04, -0.02]
2 Apgar score of 6 or less at 1 minute (not prespecified in protocol)	2	211	Risk Ratio (M-H, Fixed, 95% CI)	0.89 [0.38, 2.07]

3 Apgar score of 6 or less at 5 minutes (not prespecified in protocol)	2	211	Risk Ratio (M-H, Fixed, 95% CI)	0.60 [0.15, 2.44]
4 Need for oxygen therapy or mask ventilation of neonate	1	158	Risk Ratio (M-H, Fixed, 95% CI)	1.14 [0.49, 2.65]
5 Mean Apgar score at 1 minute	1	30	Mean Difference (IV, Fixed, 95% CI)	0.25 [-0.14, 0.64]
6 Mean Apgar score at 5 minutes	1	30	Mean Difference (IV, Fixed, 95% CI)	0.0 [0.0, 0.0]

#### Analysis I.I. Comparison | Epidural versus general anaesthesia, Outcome | Mean umbilical arterial pH.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: I Mean umbilical arterial pH

Study or subgroup	Epidural anaesthesia		General anaesthesia		Mean Difference	Mean Difference	
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Random,95% CI	IV,Random,95% CI	
I Non-urgent indication	for caesarean section						
Bengi Sener 2003	15	7.27 (0)	15	7.26 (0)		0.0 [ 0.0, 0.0 ]	
Datta 1983	10	7.31 (0.03)	10	7.32 (0.03)	+	-0.01 [ -0.04, 0.02 ]	
Dick 1992	23	7.3 (0.04)	24	7.27 (0.04)	-	0.03 [ 0.01, 0.05 ]	
Hollmen 1978	15	7.29 (0.03)	15	7.29 (0.04)	+	0.0 [ -0.03, 0.03 ]	
Pence 2002	26	7.36 (0.18)	30	7.35 (0.17)		0.01 [ -0.08, 0.10 ]	
Petropoulos 2003	72	7.28 (0.03)	80	7.29 (0.02)	+	-0.01 [ -0.02, 0.00 ]	
Yegin 2003	31	7.27 (0.08)	31	7.25 (0.07)	-	0.02 [ -0.02, 0.06 ]	
Subtotal (95% CI)	192		205		•	0.00 [ -0.01, 0.02 ]	
Heterogeneity: $Tau^2 = 0$ .	00; Chi <sup>2</sup> = 12.50, df = 5	$(P = 0.03); I^2 =$	=60%				
Test for overall effect: Z	= 0.46 (P = 0.64)						
2 Urgent indication for ca	aesarean section: severe	pre-eclampsia					
Wallace 1995	31	7.26 (0.06)	26	7.3 (0.05)	-	-0.04 [ -0.07, -0.01 ]	
Subtotal (95% CI)	31		26		•	-0.04 [ -0.07, -0.01 ]	
Heterogeneity: not applie	cable						
Test for overall effect: Z	= 2.75 (P = 0.0060)						
Total (95% CI)	223		231		•	0.00 [ -0.02, 0.01 ]	
Heterogeneity: Tau <sup>2</sup> = 0.00; Chi <sup>2</sup> = 18.14, df = 6 (P = 0.01); I <sup>2</sup> = 67%							
Test for overall effect: $Z = 0.25$ (P = 0.80)							
Test for subgroup differen	nces: $Chi^2 = 6.86$ , $df = 1$	$(P = 0.01), 1^2 =$	=85%				
						1	

-0.5 -0.25 0 0.25 0.5

Favours general Favours epidural

#### Analysis I.2. Comparison I Epidural versus general anaesthesia, Outcome 2 Mean umbilical venous pH.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 2 Mean umbilical venous pH

Study or subgroup	Epidural anaesthesia N	Mean(SD)	General anaesthesia N	Mean(SD)	Mean Difference IV,Random,95% CI	Weight	Mean Difference IV,Random,95% Cl
Datta 1983	10	7.37 (0.06)	10	7.36 (0.03)	-	7.2 %	0.01 [ -0.03, 0.05 ]
Dick 1992	23	7.3 (0.07)	24	7.3 (0.04)	+	9.9 %	0.0 [ -0.03, 0.03 ]
Hollmen 1978	15	7.34 (0.03)	15	7.33 (0.04)	+	13.2 %	0.01 [ -0.02, 0.04 ]
Lertakyamanee 1999	9 120	7.31 (0.06)	103	7.29 (0.05)	-	19.8 %	0.02 [ 0.01, 0.03 ]
Mahajan 1992	30	7.34 (0.04)	30	7.33 (0.05)	+	14.5 %	0.01 [ -0.01, 0.03 ]
Yegin 2003	31	7.33 (0.03)	31	7.31 (0.04)	-	17.7 %	0.02 [ 0.00, 0.04 ]
Yentur 2009	33	7.31 (0.05)	30	7.33 (0.01)	-	17.8 %	-0.02 [ -0.04, 0.00 ]
Total (95% CI) Heterogeneity: Tau <sup>2</sup> = ( Test for overall effect: Z Test for subgroup differe	<b>262</b> 0.00; Chi <sup>2</sup> = 14.64, df = = 1.11 (P = 0.27) ences: Not applicable	= 6 (P = 0.02);	<b>243</b> I <sup>2</sup> =59%			100.0 % (	0.01 [ -0.01, 0.02 ]

-0.5 -0.25 0 0.25 0.5 Favours general

Favours epidural

#### Analysis I.3. Comparison I Epidural versus general anaesthesia, Outcome 3 Apgar score of 4 or less at I minute.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 3 Apgar score of 4 or less at 1 minute

Study or subgroup	Epidural anaesthesia	General anaesthesia	F	Risk Ratio		Risk Ratio
	n/N	n/N	M-H,Fi>	ked,95% Cl		M-H,Fixed,95% Cl
Dick 1992	0/23	3/24			100.0 %	0.15 [ 0.01, 2.73 ]
Total (95% CI)	23	24		-	100.0 %	0.15 [ 0.01, 2.73 ]
Total events: 0 (Epidura	anaesthesia), 3 (General ar	naesthesia)				
Heterogeneity: not appl	icable					
Test for overall effect: Z	= 1.28 (P = 0.20)					
Test for subgroup differe	ences: Not applicable					
			0.001 0.01 0.1	1 10 100 1000		
			Favours epidural	Favours general		

#### Analysis I.4. Comparison I Epidural versus general anaesthesia, Outcome 4 Apgar score of 4 or less at 5 minutes.

Risk Ratio
ed,95% CI
).01, 8.11]
, <b>8.11</b> ]
Ris æd, 0.01

# Analysis 1.5. Comparison I Epidural versus general anaesthesia, Outcome 5 Apgar score of 6 or less at I minute (not prespecified in protocol).

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 5 Apgar score of 6 or less at 1 minute (not prespecified in protocol)

Study or subgroup	Epidural anaesthesia	General anaesthesia	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl		M-H,Fixed,95% CI
I Non-urgent indication fo	r caesarean section				
Petropoulos 2003	3/72	5/80		46.6 %	0.67 [ 0.17, 2.69 ]
Subtotal (95% CI)	72	80	-	46.6 %	0.67 [ 0.17, 2.69 ]
Total events: 3 (Epidural an	aesthesia), 5 (General anae	sthesia)			
Heterogeneity: not applical	ble				
Test for overall effect: Z =	0.57 (P = 0.57)				
2 Urgent indication for cae	sarean section: severe pre-e	eclampsia			
Wallace 1995	3/31	5/26		53.4 %	0.50 [ 0.13, 1.91 ]
Subtotal (95% CI)	31	26	-	53.4 %	0.50 [ 0.13, 1.91 ]
Total events: 3 (Epidural an	aesthesia), 5 (General anae	sthesia)			
Heterogeneity: not applical	ble				
Test for overall effect: Z =	1.01 (P = 0.31)				
Total (95% CI)	103	106	•	100.0 %	0.58 [ 0.22, 1.52 ]
Total events: 6 (Epidural an	aesthesia), 10 (General ana	esthesia)			
Heterogeneity: $Chi^2 = 0.08$	B, df = 1 (P = 0.77); $l^2 = 0.0^{\circ}$	%			
Test for overall effect: $Z =$	I.II (P = 0.27)				
Test for subgroup differenc	es: $Chi^2 = 0.08$ , $df = 1$ (P =	= 0.78), I <sup>2</sup> =0.0%			
			0.001 0.01 0.1 1 10 100 1000	1	
			Favours epidural Favours general		

# Analysis 1.6. Comparison I Epidural versus general anaesthesia, Outcome 6 Apgar score of 6 or less at 5 minutes (not prespecified in protocol).

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 6 Apgar score of 6 or less at 5 minutes (not prespecified in protocol)

Study or subgroup	Epidural anaesthesia n/N	General anaesthesia n/N	Risk M-H,Fixec	< Ratio I,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
I Non-urgent indication for	caesarean section					
Petropoulos 2003	2/72	3/80		-	51.2 %	0.74 [ 0.13, 4.31 ]
Subtotal (95% CI)	72	80	-	- 51	1.2 %	0.74 [ 0.13, 4.31 ]
Total events: 2 (Epidural ana Heterogeneity: not applicab Test for overall effect: Z = 0 2 Urgent indication for caes	aesthesia), 3 (General anaes le 0.33 (P = 0.74) arean section: severe pre-ee	thesia) clampsia				
Wallace 1995	0/31	2/26		-	48.8 %	0.17 [ 0.01, 3.37 ]
Subtotal (95% CI)	31	26		48	8.8 %	0.17 [ 0.01, 3.37 ]
Total events: 0 (Epidural ana Heterogeneity: not applicab Test for overall effect: Z = 1	aesthesia), 2 (General anaes le .17 (P = 0.24)	thesia)				
Total (95% CI)	103	106	•	100	0.0 %	0.46 [ 0.11, 1.95 ]
Test for overall effect: Z = 1 Test for subgroup difference	1.05 (P = 0.29) es: Chi <sup>2</sup> = 0.70, df = 1 (P =	0.40), l <sup>2</sup> =0.0%				
			0.001 0.01 0.1	10 100 1000		
			Favours epidural	Favours general		

#### Analysis I.7. Comparison I Epidural versus general anaesthesia, Outcome 7 Mean Apgar score at I minute.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 7 Mean Apgar score at 1 minute

Study or subgroup	Epidural anaesthesia N	Mean(SD)	General anaesthesia N	Mean(SD)	Mean Difference IV,Random,95% Cl	Weight	Mean Difference IV,Random,95% CI
Hodgkinson 1980	10	6.8 (2.9)	10	5.7 (2.31)		7.3 %	1.10 [ -1.20, 3.40 ]
Lertakyamanee 1999	120	8.3 (1.9)	103	6.7 (2.8)	-	22.1 %	1.60 [ 0.96, 2.24 ]
Mathur 2002	20	6.7 (1.2)	20	6.55 (1.32)	+	20.4 %	0.15 [ -0.63, 0.93 ]
Yegin 2003	31	7.38 (0.55)	31	7.19 (0.7)	-	25.4 %	0.19 [ -0.12, 0.50 ]
Yentur 2009	33	8 (0.9)	30	8.5 (0.7)	-	24.7 %	-0.50 [ -0.90, -0.10 ]
<b>Total (95% CI)</b> Heterogeneity: Tau <sup>2</sup> = ( Test for overall effect: Z Test for subgroup differe	<b>214</b> 0.52; $Chi^2 = 30.85$ , df = = 1.05 (P = 0.29) ences: Not applicable	= 4 (P<0.0000	<b>194</b> I); I <sup>2</sup> =87%		•	100.0 % (	0.39 [ -0.34, 1.12 ]

0 5 -10 -5 Favours general Favours epidural

10

#### Analysis I.8. Comparison I Epidural versus general anaesthesia, Outcome 8 Mean Apgar score at 5 minutes.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 8 Mean Apgar score at 5 minutes

Study or subgroup	Epidural anaesthesia N	Mean(SD)	General anaesthesia N	Mean(SD)	Mean Difference IV,Random,95% Cl	Weight	Difference IV,Random,95% CI
Hodgkinson 1980	10	7.9 (2.51)	10	8.4 (1.07)		3.9 %	-0.50 [ -2.19, 1.19 ]
Lertakyamanee 1999	120	9.7 (0.9)	103	9.2 (1.6)	-	29.0 %	0.50 [ 0.15, 0.85 ]
Yegin 2003	31	9.87 (0.42)	31	9.54 (0.67)	-	32.3 %	0.33 [ 0.05, 0.61 ]
Yentur 2009	33	9.6 (0.4)	30	9.7 (0.5)	•	34.8 %	-0.10 [ -0.33, 0.13 ]
<b>Total (95% CI)</b> Heterogeneity: Tau <sup>2</sup> = ( Test for overall effect: Z Test for subgroup differ	<b>194</b> 0.08; Chi <sup>2</sup> = 10.79, df = = 1.11 (P = 0.27) ences: Not applicable	= 3 (P = 0.01);	174   <sup>2</sup> =72%			100.0 %	0.20 [ -0.15, 0.55 ]

-10 -5 0 5 10 Favours general Favours epidural



### Analysis I.9. Comparison I Epidural versus general anaesthesia, Outcome 9 Neonatal neurologic and adaptive capacity score at 2-4 hours.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

-

-

Outcome: 9 Neonatal neurologic and adaptive capacity score at 2-4 hours

Study or subgroup	Epidural anaesthesia		General anaesthesia		Dif	Mean ference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Rano	dom,95% Cl		IV,Random,95% CI
Bengi Sener 2003	15	19.07 (0.89)	15	15.2 (2.09)		-	49.6 %	3.87 [ 2.72, 5.02 ]
Lertakyamanee 1999	9 120	34.9 (4.2)	103	34.4 (3.4)		-	50.4 %	0.50 [ -0.50, 1.50 ]
Total (95% CI)	135		118			-	100.0 % 2	.17 [ -1.13, 5.47 ]
Heterogeneity: $Tau^2 = \frac{1}{2}$	5.38; Chi <sup>2</sup> = 18.83, df =	= I (P = 0.000	01); I <sup>2</sup> =95%					
Test for overall effect: Z	E = 1.29 (P = 0.20)							
Test for subgroup differ	ences: Not applicable							
				i	1		1	
				-10	D -5	0 5	10	
				Favo	ours general	Favours ep	pidural	

### Analysis 1.10. Comparison I Epidural versus general anaesthesia, Outcome 10 Amount of blood transfusion received in units (not prespecified in protocol).

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 10 Amount of blood transfusion received in units (not prespecified in protocol)

Study or subgroup	Epidural anaesthesia N	Mean(SD)	General anaesthesia N	Mean(SD)	Diffe IV,Fixe	Mean erence d,95%	Weigh	Mean Difference IV,Fixed,95% Cl
Hong 2002	13	0.38 (0.9)	12	1.08 (1.6)	+-		100.0	% -0.70 [ -1.73, 0.33 ]
Total (95% CI) Heterogeneity: not ap Test for overall effect: Test for subgroup diff	13 oplicable z Z = 1.33 (P = 0.18) erences: Not applicable	e	12		•		100.0 %	% -0.70 [ -1.73, 0.33 ]
				-10 Favou	-5 ( rs epidural	) Favo	5 10 ours general	

### Analysis 1.11. Comparison I Epidural versus general anaesthesia, Outcome 11 Number who received postoperative blood transfusion (not prespecified in protocol).

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: II Number who received postoperative blood transfusion (not prespecified in protocol)

Study or subgroup	dy or subgroup Epidural anaesthesia General anaesthesia Risk Ratio		Weight	Risk Ratio M-	
	n/N	n/N	H,Random,95% Cl		H,Random,95% Cl
Hong 2002	2/13	5/12		39.7 %	0.37 [ 0.09, 1.56 ]
Lertakyamanee 1999	/  7	8/114	-	60.3 %	1.34 [ 0.56, 3.21 ]
Total (95% CI)	130	126	•	100.0 %	0.80 [ 0.23, 2.76 ]
Total events: 13 (Epidural a	anaesthesia), 13 (General anae	sthesia)			
Heterogeneity: $Tau^2 = 0.4$	6; Chi <sup>2</sup> = 2.25, df = 1 (P = 0.1	3); I <sup>2</sup> =56%			
Test for overall effect: Z =	0.35 (P = 0.73)				
Test for subgroup difference	ces: Not applicable				
				1	
			0.001 0.01 0.1 1 10 100 10	000	

Favours epidural	Favours general



Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 12 Maternal estimated blood loss in mL

Study or subgroup	Epidural anaesthesia N	Mean(SD)	General anaesthesia N	Mean(SD)	Di <sup>r</sup> IV,Fixe	Std. Mean fference ed,95% Cl	Weight	Std. Mean Difference IV,Fixed,95% CI
Hong 2002	13	1418 (996)	12	1623 (775)			9.8 %	-0.22 [ -1.01, 0.57 ]
Lertakyamanee 1999	)   7	748.2 (363.5)	114	873.6 (403.1)	-		90.2 %	-0.33 [ -0.59, -0.07 ]
<b>Total (95% CI)</b> Heterogeneity: Chi <sup>2</sup> = ( Test for overall effect: Z Test for subgroup differ	130 0.06, df = 1 (P = 0.80); = $2.51$ (P = 0.012) ences: Not applicable	; l <sup>2</sup> =0.0%	126		•		100.0 % -0	0.32 [ -0.56, -0.07 ]
				-2	-1	0 I	2	
				Favor	urs epidural	Favours ge	eneral	

Regional versus general anaesthesia for caesarean section (Review)

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

# Analysis 1.13. Comparison I Epidural versus general anaesthesia, Outcome 13 Difference between pre and postoperative haematocrit (%).

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 13 Difference between pre and postoperative haematocrit (%)

Study or subgroup	Epidural anaesthesia N	Mean(SD)	General anaesthesia N	Mean(SD)	M Differe	lean ence 95% CI	Weight	Mean Difference IVEixed 95% Cl
Lertakyamanee 1999	9 117	-3.4 (4.8)	4	-5.1 (4.7)		+	100.0 %	1.70 [ 0.47, 2.93 ]
<b>Total (95% CI)</b> Heterogeneity: not app Test for overall effect: Z	<b>117</b> licable Z = 2.72 (P = 0.0065)		114			•	100.0 % 1.	70 [ 0.47, 2.93 ]
Test for subgroup differ	ences: Not applicable							
				-10 Favour	-5 0	5 IC	D	

# Analysis 1.14. Comparison I Epidural versus general anaesthesia, Outcome 14 Satisfaction score on visual analogue scale.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 14 Satisfaction score on visual analogue scale



#### Analysis 1.15. Comparison I Epidural versus general anaesthesia, Outcome 15 Number who would prefer the same technique again.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 15 Number who would prefer the same technique again

Study or subgroup	Epidural anaesthesia n/N	General anaesthesia n/N	Risk M-H,Fixed,	Ratio ,95% CI	Weight	Risk Ratio M-H,Fixed,95% Cl
Lertakyamanee 1999	65/120	70/103			100.0 %	0.80 [ 0.65, 0.98 ]
Total (95% CI)	120	103	•		100.0 %	0.80 [ 0.65, 0.98 ]
Total events: 65 (Epidural a	anaesthesia), 70 (General a	naesthesia)				
Heterogeneity: not applica	ble					
Test for overall effect: Z =	2.10 (P = 0.035)					
Test for subgroup difference	es: Not applicable					
			0.1 0.2 0.5 1	2 5 10		
			Favours general F	avours epidural		

### Analysis 1.16. Comparison I Epidural versus general anaesthesia, Outcome 16 Neonatal neurologic and adaptive capacity score of < 35 at 15 minutes.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 16 Neonatal neurologic and adaptive capacity score of < 35 at 15 minutes

Study or subgroup	Epidural anaesthesia	General anaesthesia	F	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fix	xed,95% Cl		M-H,Fixed,95% Cl
Mahajan 1992	17/30	18/30	-	-	100.0 %	0.94 [ 0.62, 1.45 ]
Total (95% CI)	30	30	•	-	100.0 %	0.94 [ 0.62, 1.45 ]
Total events: 17 (Epidur	al anaesthesia), 18 (General a	anaesthesia)				
Heterogeneity: not appl	icable					
Test for overall effect: Z	= 0.26 (P = 0.79)					
Test for subgroup differe	ences: Not applicable					
			0.1 0.2 0.5	2 5 10		
			Eavours epidural	Eavours general		

### Analysis 1.17. Comparison I Epidural versus general anaesthesia, Outcome 17 Neonatal neurologic and adaptive capacity score of < 35 at 2 hours.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 17 Neonatal neurologic and adaptive capacity score of < 35 at 2 hours

Study or subgroup	Epidural anaesthesia n/N	General anaesthesia n/N	I M-H,Fix	Risk Ratio xed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
Mahajan 1992	6/30	9/30		-	100.0 %	0.67 [ 0.27, 1.64 ]
Total (95% CI)	30	30	-	-	100.0 %	0.67 [ 0.27, 1.64 ]
Total events: 6 (Epidura	l anaesthesia), 9 (General an	aesthesia)				
Heterogeneity: not app	licable					
Test for overall effect: Z	E = 0.88 (P = 0.38)					
Test for subgroup differ	ences: Not applicable					
			0.1 0.2 0.5	2 5 10		
			Favours epidural	Favours general		

#### Analysis 1.18. Comparison | Epidural versus general anaesthesia, Outcome 18 Adverse events.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 18 Adverse events

Study or subgroup	Epidural	General	ſ	Risk Ratio M-	Weight	Risk Ratio M-
	n/N	n/N	H,Rar	ndom,95% Cl		H,Random,959 Cl
l Headache						
Hodgkinson 1980	6/10	5/10	ł	-	100.0 %	1.20 [ 0.54, 2.67 ]
Subtotal (95% CI)	10	10		•	100.0 %	1.20 [ 0.54, 2.67 ]
Total events: 6 (Epidural), 5 (C	General)					
Heterogeneity: not applicable						
lest for overall effect: $\angle = 0.4$	-5 (P = 0.66)					
2 Epigastric pain Hodgkinson 1980	0/10	3/10		_	100.0 %	0.14 [ 0.01, 2.45 ]
Subtatal (05% CI)	10	10			100 0 %	01/[001 2/5]
Total events: 0 (Epidural) 3 (C	General)	10	_		100.0 /0	0.14 [ 0.01, 2.49 ]
Heterogeneity: not applicable	Serieraly					
Test for overall effect: $Z = 1.3$	4 (P = 0.18)					
3 Blurred vision						
Hodgkinson 1980	4/10	2/10	-	<b>-</b>	100.0 %	2.00 [ 0.47, 8.56 ]
Subtotal (95% CI)	10	10	-	•	100.0 %	2.00 [ 0.47, 8.56 ]
Total events: 4 (Epidural), 2 (C	General)					
Heterogeneity: not applicable						
Test for overall effect: $Z = 0.9$	3 (P = 0.35)					
4 Convulsion						
Hodgkinson 1980	2/10	2/10			100.0 %	1.00 [ 0.17, 5.77 ]
Subtotal (95% CI)	10	10		-	100.0 %	1.00 [ 0.17, 5.77 ]
Total events: 2 (Epidural), 2 (C	General)					
Heterogeneity: not applicable						
lest for overall effect: $\angle = 0.0$	P = 1.0					
Bengi Sener 2003	9/15	14/15		•	35.2 %	0.64 [ 0.42, 0.99 ]
Hong 2002	5/13	3/12	-	-	31.6 %	1.54 [ 0.46, 5.09 ]
Lertakvamanee 1999	33/117	5/114			33.3 %	6.43 [ 2.60, 15.89 ]
Subtotal (95% CI)	1/5	1/1			100.0.%	
Total events: 47 (Epidural) 22	(General)	141			100.0 /0	1.02 [ 0.20, 12.00 ]
Heterogeneity: $Tau^2 = 2.76$ : C	$Chi^2 = 34.44. df = 2$	(P<0.00001);   <sup>2</sup> =	-94%			
		(				
			0.001 0.01 0.1	10 100 1000		
			Favours epidural	Favours general		
						(Continued )

Epidural	General	Risk Ratio	Weight	( Continued) Risk Ratio
n/N	n/N	M- H,Random,95% Cl		M- H,Random,959 Cl
(P = 0.55)				
2/15	9/15		35.9 %	0.22 [ 0.06, 0.86 ]
21/117	8/114	-	41.6 %	2.56 [ 1.18, 5.54 ]
1/20	1/20	<b>_</b>	22.5 %	1.00 [ 0.07, 14.90 ]
152	149	-	100.0 %	0.86 [ 0.14, 5.25 ]
(General) hi <sup>2</sup> = 9.5 I, df = 2 ( h (P = 0.87)	P = 0.01); I <sup>2</sup> =79%			
4/13	0/12		100.0 %	8.36 [ 0.50, 140.56 ]
<b>13</b> eneral) (P = 0.14)	12		100.0 %	8.36 [ 0.50, 140.56 ]
1/15	8/15		100.0 %	0.13 [ 0.02, 0.88 ]
15 eneral) (P = 0.037) 0/15	4/15	-	100.0 %	0.13 [ 0.02, 0.88 ]
15	15	-	100.0 %	0.11 [ 0.01, 1.90 ]
eneral) . (P = 0.13)	15			
3/15	2/15		100.0 %	1.50 [ 0.29, 7.73 ]
<b>15</b> eneral) (P = 0.63)	15		100.0 %	1.50 [ 0.29, 7.73 ]
	(	0.001 0.01 0.1 10 100 1000 Favours epidural Favours general		
	Epidural n/N (P = 0.55) 2/15 21/117 1/20 152 General) $i^2 = 9.51, df = 2 (10)$ (P = 0.87) 4/13 13 eneral) (P = 0.14) 1/15 15 eneral) (P = 0.037) 0/15 15 eneral) (P = 0.13) 3/15 15 eneral) (P = 0.63)	Epidural       General $n/N$ $n/N$ (P = 0.55) $2/15$ $2/15$ $9/15$ $21/117$ $8/114$ $1/20$ $1/20$ $152$ $149$ General) $i^2 = 9.51$ , df = 2 (P = 0.01); $I^2 = 79\%$ (P = 0.87) $4/13$ $0/12$ $13$ $12$ eneral) $4/13$ $(P = 0.14)$ $1/15$ $15$ $15$ eneral) $(P = 0.037)$ $0/15$ $4/15$ $15$ $15$ eneral) $(P = 0.13)$ $3/15$ $2/15$ $15$ $15$ eneral) $(P = 0.63)$	Epidural       General       Risk Ratio $n/N$ $n/N$ $n/N$ $HRandom,95\%$ $(P = 0.55)$ $2/15$ $9/15$ $\bullet$ $21/117$ $8/114$ $\bullet$ $\bullet$ $1/20$ $1/20$ $\bullet$ $\bullet$ $152$ $149$ $\bullet$ $\bullet$ General) $i^2 = 9.51$ , df = 2 (P = 0.01); l^2 = 79% $\bullet$ $\bullet$ $(P = 0.87)$ $4/13$ $0/12$ $\bullet$ $\bullet$ $4/13$ $0/12$ $\bullet$ $\bullet$ $\bullet$ $(P = 0.14)$ $1/15$ $8/15$ $\bullet$ $\bullet$ $(P = 0.14)$ $1/15$ $8/15$ $\bullet$ $\bullet$ $(P = 0.13)$ $0/15$ $4/15$ $\bullet$ $\bullet$ $(P = 0.13)$ $3/15$ $2/15$ $\bullet$ $\bullet$ $(P = 0.63)$ $\bullet$ $\bullet$ $\bullet$ $\bullet$ $0.001 0.01 0.1$ $I0 100 1000$ Favours gendral       Favours gendral	Epidural         General         Risk Ratio M. HRandom,95% C         Weight           (P = 0.55)         2/15         9/15         35.9 %           21/117         8/114         41.6 %           1/20         1/20         22.5 %           152         149         100.0 %           General)         100.0 %         152           4/13         0/12         100.0 %           13         12         100.0 %           13         12         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %           15         15         100.0 %

# Analysis 1.19. Comparison I Epidural versus general anaesthesia, Outcome 19 Need for oxygen therapy or mask ventilation of the neonate.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 19 Need for oxygen therapy or mask ventilation of the neonate

Study or subgroup	Epidural anaesthesia	General anaesthesia		Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fi	xed,95% Cl		M-H,Fixed,95% Cl
Petropoulos 2003	7/72	9/80			100.0 %	0.86 [ 0.34, 2.20 ]
Total (95% CI)	72	80			100.0 %	0.86 [ 0.34, 2.20 ]
Total events: 7 (Epidural	anaesthesia), 9 (General ana	aesthesia)				
Heterogeneity: not appli	cable					
Test for overall effect: Z	= 0.31 (P = 0.76)					
Test for subgroup differe	nces: Not applicable					
			0.1 0.2 0.5	1 2 5 10		
			Favours epidural	Favours general		

### Analysis I.20. Comparison I Epidural versus general anaesthesia, Outcome 20 Maternal blood loss > 500 mL.

Review: Regional versus	general anaesthesia f	or caesarean sectio	n		
Comparison: I Epidural	versus general anaes	thesia			
Outcome: 20 Maternal b	blood loss > 500 mL				
Study or subgroup	Epidural n/N	General n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
Mathur 2002	2/20	4/20		100.0 %	0.50 [ 0.10, 2.43 ]
Total (95% CI)	20	20		100.0 %	0.50 [ 0.10, 2.43 ]
Total events: 2 (Epidural), 4 Heterogeneity: not applical Test for overall effect: Z = 1 Test for subgroup difference	ł (General) ble 0.86 (P = 0.39) es: Not applicable				
			0.01 0.1 10 100 Favours epidural Favours genera	) al	

#### Analysis 1.21. Comparison I Epidural versus general anaesthesia, Outcome 21 Mean Apgar score at 10 minutes.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 21 Mean Apgar score at 10 minutes

Study or subgroup	Epidural		General			Diff	Mean erence		Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)		IV,Fixe	ed,95% Cl			IV,Fixed,95% CI
Mathur 2002	20	8.25 (0.69)	20	8.7 (1.14)		-			100.0 %	-0.45 [ -1.03, 0.13 ]
Total (95% CI)	20		20			•			100.0 %	-0.45 [ -1.03, 0.13 ]
Heterogeneity: not app	olicable									
Test for overall effect:	Z = 1.51 (P =	0.13)								
Test for subgroup diffe	rences: Not aj	pplicable								
					-10	-5	0 5	10		
					Favours	epidural	Favou	rs general		

#### Analysis 1.22. Comparison I Epidural versus general anaesthesia, Outcome 22 Time to request postoperative analgesia in minutes.

Review: Regional versus gen	eral anaesthe	esia for caesarean s	ection					
Comparison: I Epidural vers	sus general ar	naesthesia						
Outcome: 22 Time to reque	est postopera	ative analgesia in mi	nutes					
Study or subgroup Epidural a	anaesthesia	Gen Maga (CD)	eral anaesthesia	Maar(CD)	D.	Mean Difference	Weight	Mear Difference
	IN	I*lean(SD)	IN	I*lean(SD)	IV,	FIXED,95% CI		IV,FIXED,95% C
Hong 2002	13	690 (30)	12	190 (238)			100.0 %	500.00 [ 364.36, 635.64 ]
Total (95% CI)	13		12			•	100.0 %	500.00 [ 364.36, 635.64 ]
Heterogeneity: not applicable								
Test for overall effect: Z = 7.22	P < 0.0000	)))						
Test for subgroup differences: N	Not applicabl	e						
				-1000	-500	0 500	1000	
				Favours	s epidura	I Favour	s general	

# Analysis 1.23. Comparison I Epidural versus general anaesthesia, Outcome 23 Intraoperative pain score on visual analogue scale.

Review: Regional versus general anaesthesia for caesarean section

Comparison: I Epidural versus general anaesthesia

Outcome: 23 Intraoperative pain score on visual analogue scale



#### Analysis 2.1. Comparison 2 Spinal versus general anaesthesia, Outcome 1 Umbilical arterial pH.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: I Umbilical arterial pH

Study or subgroup	Spinal anaesthesia N	Mean(SD)	General anaesthesia N	Mean(SD)	Diffe IV,Fixed	Mean rence 1,95% Cl	Weight	Mean Difference IV,Fixed,95% CI
Akyol 2006	30	7.31 (0.02)	32	7.31 (0.02)	-	ŀ	39.9 %	0.00 [ -0.01, 0.01 ]
Datta 1983	10	7.31 (0.06)	10	7.32 (0.03)			2.3 %	-0.01 [ -0.05, 0.03 ]
Kavak 2001	46	7.24 (0.01)	38	7.25 (0.08)		_	6.0 %	-0.01 [ -0.04, 0.02 ]
Mahajan 1992	30	7.28 (0.02)	30	7.28 (0.04)		_	15.5 %	0.0 [ -0.02, 0.02 ]
Mancuso 2010	90	7.27 (0.04)	81	7.28 (0.03)	-		35.7 %	-0.01 [ -0.02, 0.00 ]
Moslemi 2007	32	7.275 (0.154)	30	7.26 (0.174)			- 0.6 %	0.02 [ -0.07, 0.10 ]
Total (95% CI) Heterogeneity: Chi <sup>2</sup> Test for overall effect Test for subgroup dif	<b>238</b> = 3.38, df = 5 (P = :: Z = 1.09 (P = 0.27) ferences: Not applic	0.64); I <sup>2</sup> =0.0% 7) able	221		•		100.0 %	0.00 [ -0.01, 0.00 ]
				_(	0.1 -0.05 0	0.05	0.1	

Favours general Favours spinal

#### Analysis 2.2. Comparison 2 Spinal versus general anaesthesia, Outcome 2 Umbilical venous pH.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 2 Umbilical venous pH

Study or subgroup	Spinal anaesthesia		General anaesthesia		Mean Difference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Fixed,95% CI		IV,Fixed,95% CI
Datta 1983	10	7.37 (0.03)	10	7.36 (0.03)	-	15.2 %	0.01 [ -0.02, 0.04 ]
Lertakyamanee 1999	118	7.3 (0.06)	103	7.29 (0.05)	•	50.0 %	0.01 [ 0.00, 0.02 ]
Mahajan 1992	30	7.34 (0.05)	30	7.33 (0.05)	-	16.4 %	0.01 [ -0.02, 0.04 ]
Nabhan 2009	41	7.25 (0.06)	41	7.25 (0.05)	+	18.4 %	0.0 [ -0.02, 0.02 ]
Total (95% CI)	199		184			100.0 %	0.01 [ 0.00, 0.02 ]
Heterogeneity: $Chi^2 = 0$	0.55, df = 3 (P = 0.91	); I <sup>2</sup> =0.0%					
Test for overall effect: Z	= 1.56 (P = 0.12)						
Test for subgroup differe	ences: Not applicable						

-0.5 -0.25 0 0.25 0.5 Favours general Favours spinal

# Analysis 2.3. Comparison 2 Spinal versus general anaesthesia, Outcome 3 Neonatal neurologic and adaptive capacity score at 2-4 hours.

Review: Regional vers	us general anaesthesi	a for caesarea	n section					
Comparison: 2 Spinal	versus general anaes	thesia						
Outcome: 3 Neonata	l neurologic and adap	tive capacity	score at 2-4 hours					
Study or subgroup	Spinal anaesthesia		General anaesthesia		Diffe	Mean erence	Weight	Mean Difference
	Ν	Mean(SD)	N	Mean(SD)	IV,Fixe	d,95% Cl		IV,Fixed,95% CI
Lertakyamanee 1999	118	34.8 (3.7)	103	34.4 (3.4)			100.0 %	0.40 [ -0.54, 1.34 ]
Total (95% CI)	118		103			•	100.0 %	0.40 [ -0.54, 1.34 ]
Heterogeneity: not appli	cable							
Test for overall effect: Z	= 0.84 (P = 0.40)							
Test for subgroup differe	nces: Not applicable							
							1	
				-10	-5 (	0 5	10	
				Favo	urs general	Favours sp	inal	

Regional versus general anaesthesia for caesarean section (Review)

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

# Analysis 2.4. Comparison 2 Spinal versus general anaesthesia, Outcome 4 Neonatal neurologic and adaptive capacity score of < 35 at 15 minutes.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 4 Neonatal neurologic and adaptive capacity score of < 35 at 15 minutes

Study or subgroup	ıdy or subgroup Spinal anaesthesia General anaesthesia Risk Ratio n/N n/N M-H,Fixed,95% Cl		Risk Ratio ×ed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl		
Mahajan 1992	3/30	18/30			100.0 %	0.17 [ 0.05, 0.51 ]	
Total (95% CI) Total events: 3 (Spinal a Heterogeneity: not app Test for overall effect: Z Test for subgroup diffen	<b>30</b> Inaesthesia), 18 (General an licable Z = 3.16 (P = 0.0016) ences: Not applicable	<b>30</b> aesthesia)	•		100.0 %	0.17 [ 0.05, 0.51 ]	
			<u> </u>				
			0.001 0.01 0.1 Favours spinal	I 10 100 1000 Favours general			

### Analysis 2.5. Comparison 2 Spinal versus general anaesthesia, Outcome 5 Neonatal neurologic and adaptive capacity score of < 35 at 2 hours.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

.

Outcome: 5 Neonatal neurologic and adaptive capacity score of < 35 at 2 hours

Study or subgroup	Spinal anaesthesia	General anaesthesia	I	Risk Ratio		Risk Ratio
	n/N	n/N	M-H,Fi	M-H,Fixed,95% Cl		M-H,Fixed,95% Cl
Mahajan 1992	0/30	9/30	← <mark></mark>	-	100.0 %	0.05 [ 0.00, 0.87 ]
Total (95% CI)	30	30		-	100.0 %	0.05 [ 0.00, 0.87 ]
Total events: 0 (Spinal ar	naesthesia), 9 (General ana	aesthesia)				
Heterogeneity: not appl	icable					
Test for overall effect: Z	= 2.06 (P = 0.039)					
Test for subgroup differe	ences: Not applicable					
			0.001 0.01 0.1	10 100 1000		
			Favours spinal	Favours general		

### Analysis 2.6. Comparison 2 Spinal versus general anaesthesia, Outcome 6 Neonatal neurologic and adaptive capacity score of < 35 at 24 hours.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 6 Neonatal neurologic and adaptive capacity score of < 35 at 24 hours

Study or subgroup	udy or subgroup Spinal anaesthesia General anaesthesia			Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fi>	ked,95% Cl	M-H,Fixed,95% CI
Mahajan 1992	0/30	0/30			0.0 [ 0.0, 0.0 ]
Total (95% CI)	30	30			0.0 [ 0.0, 0.0 ]
Total events: 0 (Spinal anaest	hesia), 0 (General anaesthesia)				
Heterogeneity: not applicable	2				
Test for overall effect: $Z = 0.0$	0 (P < 0.00001)				
Test for subgroup differences	: Not applicable				
			0.1 0.2 0.5	1 2 5 10	
			Favours spinal	Favours general	

Regional versus general anaesthesia for caesarean section (Review)

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

#### Analysis 2.7. Comparison 2 Spinal versus general anaesthesia, Outcome 7 Mean Apgar score at 1 minute.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 7 Mean Apgar score at 1 minute

Study or subgroup	Spinal anaesthesia		General anaesthesia		ا Differ	Mean rence	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Rando	m,95% Cl		IV,Random,95% CI
Akyol 2006	30	8.03 (0.67)	32	7.88 (0.66)	-		22.2 %	0.15 [ -0.18, 0.48 ]
Kavak 2001	46	8.86 (0.5)	38	8.7 (0.6)	-		22.9 %	0.16 [ -0.08, 0.40 ]
Lertakyamanee 1999	118	8.7 (0.6)	103	6.7 (2.8)		•	19.9 %	2.00 [ 1.45, 2.55 ]
Mathur 2002	21	6.85 (1.3)	20	6.55 (1.32)	-	F	17.0 %	0.30 [ -0.50, 1.10 ]
Moslemi 2007	32	7.66 (1.43)	30	7.57 (1.43)	+		18.1 %	0.09 [ -0.62, 0.80 ]
Total (95% CI)	247		223		•	•	100.0 %	0.54 [ -0.09, 1.16 ]
Heterogeneity: $Tau^2 = 0$	0.43; Chi <sup>2</sup> = 38.73, df	= 4 (P<0.000	01); I <sup>2</sup> =90%					
Test for overall effect: Z	= 1.69 (P = 0.091)							
Test for subgroup differe	ences: Not applicable							
				ı			L	
				-	0 -5 0	5	10	

-10 -5 0 5 10 Favours general Favours spinal
### Analysis 2.8. Comparison 2 Spinal versus general anaesthesia, Outcome 8 Mean Apgar score at 5 minutes.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 8 Mean Apgar score at 5 minutes

Study or subgroup	Spinal anaesthesia N	Mean(SD)	General anaesthesia N	Mean(SD)	Me Differer IV,Random	ean nce Weigh ,95% Cl	t Difference IV,Random,95% CI
Akyol 2006	30	7.98 (0.55)	32	7.56 (0.46)		25.3 %	6 0.42 [ 0.17, 0.67 ]
Kavak 2001	46	9.9 (0.2)	38	9.9 (0.3)	-	29.8 9	6 0.0 [ -0.11, 0.11 ]
Lertakyamanee 1999	9   8	9.8 (0.7)	103	9.2 (1.6)	-	- 22.3 9	6 0.60 [ 0.27, 0.93 ]
Moslemi 2007	32	9.41 (0.5)	30	9.4 (0.77)	+	22.6 9	6 0.01 [ -0.32, 0.34 ]
<b>Total (95% CI)</b> Heterogeneity: Tau <sup>2</sup> = ( Test for overall effect: Z Test for subgroup differe	<b>226</b> 0.07; Chi <sup>2</sup> = 18.12, dt = 1.61 (P = 0.11) ences: Not applicable	f = 3 (P = 0.00	<b>203</b> 0042); I <sup>2</sup> =83%		•	100.0 %	o 0.24 [ -0.05, 0.54 ]
				-	-4 -2 0	2 4	

Favours general Favours spinal

# Analysis 2.9. Comparison 2 Spinal versus general anaesthesia, Outcome 9 Apgar score of 6 or less at 1 minute (not prespecified in protocol).

Review: Regional versus general anaesthesia for caesarean section

#### Comparison: 2 Spinal versus general anaesthesia

Outcome: 9 Apgar score of 6 or less at 1 minute (not prespecified in protocol)

Study or subgroup	Spinal anaesthesia	General anaesthesia		Risk Ratio	Weight	Risk Ratio
			H,Ra	M- ndom,95%		H,Random,95%
	n/N	n/N		Cl		Cl
I Non-urgent indication fo	or caesarean section					
Mahajan 1992	2/30	3/30	-		18.9 %	0.67 [ 0.12, 3.71 ]
Mancuso 2010	1/90	32/81	←		16.9 %	0.03 [ 0.00, 0.20 ]
Nabhan 2009	4/4	0/41			11.3 %	9.00 [ 0.50, 161.98 ]
Turhanoglu 1999	14/30	22/30	•	•	28.7 %	0.64 [ 0.41, 0.99 ]
Subtotal (95% CI)	191	182	-		75.7 %	0.50 [ 0.07, 3.39 ]
Total events: 21 (Spinal and	aesthesia), 57 (General anae	sthesia)				
Heterogeneity: $Tau^2 = 2.98$	8; Chi <sup>2</sup> = 18.03, df = 3 (P =	0.00043); l <sup>2</sup> =83%				
Test for overall effect: $Z =$	0.71 (P = 0.48)					
2 Urgent indication for cae	esarean section: severe pre-e	clampsia				
Moslemi 2007	5/32	6/30	-	-	24.3 %	0.78 [ 0.27, 2.29 ]
Subtotal (95% CI)	32	30	•	•	24.3 %	0.78 [ 0.27, 2.29 ]
Total events: 5 (Spinal anae	esthesia), 6 (General anaesth	nesia)				
Heterogeneity: not applica	ble					
Test for overall effect: $Z =$	0.45 (P = 0.65)					
Total (95% CI)	223	212	-		100.0 %	0.54 [ 0.16, 1.84 ]
Total events: 26 (Spinal and	aesthesia), 63 (General anae	sthesia)				
Heterogeneity: $Tau^2 = 1.32$	2; Chi <sup>2</sup> = 17.46, df = 4 (P =	0.002); l <sup>2</sup> =77%				
Test for overall effect: $Z =$	0.99 (P = 0.32)					
Test for subgroup difference	tes: $Chi^2 = 0.16$ , $df = 1$ (P =	: 0.69), I <sup>2</sup> =0.0%				
			0.001 0.01 0.1	1 10 100 1000		
			Favours spinal	Favours general		

# Analysis 2.10. Comparison 2 Spinal versus general anaesthesia, Outcome 10 Apgar score of 6 or less at 5 minutes (not prespecified in protocol).

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 10 Apgar score of 6 or less at 5 minutes (not prespecified in protocol)

Study or subgroup	Spinal anaesthesia	General anaesthesia	I	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fiz	xed,95% Cl	M-H,Fixed,95% Cl
Mahajan 1992	0/30	0/30			0.0 [ 0.0, 0.0 ]
Mancuso 2010	0/90	0/81			0.0 [ 0.0, 0.0 ]
Turhanoglu 1999	5/30	9/30			0.56 [ 0.21, 1.46 ]
Total (95% CI)	150	141			0.56 [ 0.21, 1.46 ]
Total events: 5 (Spinal anae	sthesia), 9 (General anaesthesia)				
Heterogeneity: $Chi^2 = 0.0$ ,	df = 0 (P = 1.00); l <sup>2</sup> =0.0%				
Test for overall effect: $Z =$	1.19 (P = 0.23)				
Test for subgroup difference	es: Not applicable				
			0.1 0.2 0.5	1 2 5 10	
			Favours spinal	Favours general	

# Analysis 2.11. Comparison 2 Spinal versus general anaesthesia, Outcome 11 Maternal estimated blood loss in mL.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: II Maternal estimated blood loss in mL

-

-

Study or subgroup	Favours spinal N	Mean(SD)	General anaesthesia N	Mean(SD)		Dif IV,Fixe	Std. Mean ference d,95% Cl	Weight	Std. Mean Difference IV,Fixed,95% Cl
Dyer 2003	35	394 (64)	35	446 (126)		-		25.5 %	-0.51 [ -0.99, -0.04 ]
Lertakyamanee 1999	95	648 (312)	4	873.6 (403.1)		++-		74.5 %	-0.62 [ -0.90, -0.34 ]
<b>Total (95% CI)</b> Heterogeneity: $Chi^2 = 0$	<b>130</b> .13, df = 1 (P = 0	0.72); I <sup>2</sup> =0.0%	149			•		100.0 %	-0.59 [ -0.83, -0.35 ]
Test for overall effect: Z	= 4.81 (P < 0.00)	001)							
Test for subgroup differe	nces: Not applica	able							
					1			1	
				-	2 -	.  (	)	2	

Favours spinal Favours general

## Analysis 2.12. Comparison 2 Spinal versus general anaesthesia, Outcome 12 Difference between pre and postoperative haematocrit.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 12 Difference between pre and postoperative haematocrit

Study or subgroup	Spinal anaesthesia		General anaesthesia		D	Mean Difference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,F	ixed,95% Cl		IV,Fixed,95% CI
Lertakyamanee 1999	95	-2 (5.3)	114	-5.1 (4.7)		-	100.0 %	3.10 [ 1.73, 4.47 ]
Total (95% CI)	95		114			•	100.0 %	3.10 [ 1.73, 4.47 ]
Heterogeneity: not appl	icable							
Test for overall effect: Z	= 4.43 (P < 0.0000 I	)						
Test for subgroup differe	ences: Not applicable							
				-10	-5	0 5	10	
				Favo	urs general	Favours s	pinal	

## Analysis 2.13. Comparison 2 Spinal versus general anaesthesia, Outcome 13 Number who received postoperative blood transfusion (not prespecified in protocol).

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 13 Number who received postoperative blood transfusion (not prespecified in protocol)

Study or subgroup	Spinal anaesthesia	General anaesthesia	F	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fi>	ed,95% Cl		M-H,Fixed,95% Cl
Lertakyamanee 1999	2/95	8/114		_	100.0 %	0.30 [ 0.07, 1.38 ]
Total (95% CI)	95	114	•		100.0 %	0.30 [ 0.07, 1.38 ]
Total events: 2 (Spinal anae	esthesia), 8 (General anaes	thesia)				
Heterogeneity: not applica	ble					
Test for overall effect: $Z =$	I.55 (P = 0.12)					
Test for subgroup difference	es: Not applicable					
			0.001 0.01 0.1	1 10 100 1000		
			Favours spinal	Favours general		

## Analysis 2.14. Comparison 2 Spinal versus general anaesthesia, Outcome 14 Number who would prefer the same technique again.

Review: Regional versus general anaesthesia for caesarean section										
Comparison: 2 Spinal ver	rsus general anaesthesia									
Outcome: 14 Number w	ho would prefer the same	e technique again								
Study or subgroup	Spinal anaesthesia n/N	General anaesthesia n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl					
Lertakyamanee 1999	64/118	70/103	-	100.0 %	0.80 [ 0.65, 0.99 ]					
Total (95% CI)	118	103	•	100.0 %	0.80 [ 0.65, 0.99 ]					
Total events: 64 (Spinal ana	esthesia), 70 (General ana	esthesia)								
Heterogeneity: not applicab	ble									
Test for overall effect: $Z = 2$	2.08 (P = 0.037)									
Test for subgroup difference	es: Not applicable									
			<u> </u>							
			0.1 0.2 0.5 2 5 10							
			Favours general Favours spinal							

# Analysis 2.15. Comparison 2 Spinal versus general anaesthesia, Outcome 15 Satisfaction score on visual analogue scale.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 15 Satisfaction score on visual analogue scale



## Analysis 2.16. Comparison 2 Spinal versus general anaesthesia, Outcome 16 Adverse events.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 16 Adverse events

Study or subgroup	Spinal anaesthesia	General anaesthesia	F	Risk Ratio M- H Pandom 95%		Risk Ratio
	n/N	n/N	H,Kan	Cl		H,Kandom,95% Cl
I Nausea						
Lertakyamanee 1999	49/95	5/114		-	52.0 %	.76 [ 4.88, 28.32 ]
Moslemi 2007	3/30	4/30		-	48.0 %	0.75 [ 0.18, 3.07 ]
Subtotal (95% CI)	125	144			100.0 %	3.14 [ 0.21, 48.07 ]
Total events: 52 (Spinal ana Heterogeneity: Tau <sup>2</sup> = $3.52$ Test for overall effect: Z = 0 2 Vomiting	esthesia), 9 (General anae ; Chi <sup>2</sup> = 10.82, df = 1 (P = 0.82 (P = 0.41)	sthesia) = 0.001); I <sup>2</sup> =91%				
Lertakyamanee 1999	33/95	8/114			47.5 %	4.95 [ 2.40, 10.20 ]
Mathur 2002	2/21	1/20			25.6 %	1.90 [ 0.19, 19.40 ]
Moslemi 2007	1/30	3/30			26.9 %	0.33 [ 0.04, 3.03 ]
Subtotal (95% CI) Total events: 36 (Spinal ana Heterogeneity: Tau <sup>2</sup> = $1.34$ Test for overall effect: Z = 0	<b>146</b> esthesia), 12 (General ana ; Chi <sup>2</sup> = 5.52, df = 2 (P = 0.75 (P = 0.45)	<b>164</b> esthesia) 0.06); I <sup>2</sup> =64%	-	-	100.0 %	1.88 [ 0.36, 9.69 ]
			Favours spinal	Favours general		
			·	0		

### Analysis 2.17. Comparison 2 Spinal versus general anaesthesia, Outcome 17 Maternal blood loss > 500 mL.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 17 Maternal blood loss > 500 mL

M-H,Fixed,95% CI
0.24 [ 0.03, 1.95 ]
0.24 [ 0.03, 1.95 ]

## Analysis 2.18. Comparison 2 Spinal versus general anaesthesia, Outcome 18 Mean Apgar score at 10 minutes.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 18 Mean Apgar score at 10 minutes

Study or subgroup	Spinal anaesthesia N	Mean(SD)	General anaesthesia N	Mean(SD)	Diff IV,Fixe	Mean erence ed,95% Cl	Weight	Mean Difference IV,Fixed,95% Cl
Mathur 2002	21	8.55 (I)	20	8.7 (1.14)		-	100.0 %	-0.15 [ -0.81, 0.51 ]
Total (95% CI) Heterogeneity: not a Test for overall effect	<b>21</b> applicable t: Z = 0.45 (P = 0.65)		20			•	100.0 %	-0.15 [ -0.81, 0.51 ]
Test for subgroup dif	ferences: Not applical	ble						
				-10 Favo	-5 urs spinal	0 5 Favours gen	IO eral	

Regional versus general anaesthesia for caesarean section (Review)

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

## Analysis 2.19. Comparison 2 Spinal versus general anaesthesia, Outcome 19 Time to request postoperative analgesia in minutes.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 19 Time to request postoperative analgesia in minutes

Study or subgroup Spinal anaesthes	ia	General anaesthesia		Diff	Mean erence	Weight	Mean Difference
	N Mean(SE	) N	Mean(SD)	IV,Fixe	ed,95% Cl		IV,Fixed,95% CI
Turhanoglu 1999	0 106.6 (20.8	30	8.8 (2.9)		+	100.0 %	97.80 [ 90.28, 105.32 ]
Total (95% CI) 3	0	30			•	100.0 %	97.80 [ 90.28, 105.32 ]
Heterogeneity: not applicable							
Test for overall effect: $Z = 25.51$ (P <	0.00001)						
Test for subgroup differences: Not ap	olicable						
			-200	0 -100	0 100	200	
			Favou	urs General	Favours	Spinal	

## Analysis 2.20. Comparison 2 Spinal versus general anaesthesia, Outcome 20 Neonatal need for oxygen by mask or intubation.

Review: Regional vers	us general anaesthesia for o	caesarean section				
Comparison: 2 Spinal	versus general anaesthesia					
Outcome: 20 Neonat	al need for oxygen by mas	k or intubation				
Study or subgroup	Spinal anaesthesia	General anaesthesia	F	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fix	M-H,Fixed,95% Cl		M-H,Fixed,95% Cl
Mancuso 2010	0/90	12/81	← <mark>-</mark>		100.0 %	0.04 [ 0.00, 0.60 ]
Total (95% CI)	90	81			100.0 %	0.04 [ 0.00, 0.60 ]
Total events: 0 (Spinal ar	naesthesia), 12 (General an	aesthesia)				
Heterogeneity: not appli	cable					
Test for overall effect: Z	= 2.32 (P = 0.021)					
Test for subgroup differe	ences: Not applicable					
			0.001 0.01 0.1	1 10 100 1000		
			Favours spinal	Favours general		

# Analysis 2.21. Comparison 2 Spinal versus general anaesthesia, Outcome 21 Intraoperative pain score on visual analogue scale.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 2 Spinal versus general anaesthesia

Outcome: 21 Intraoperative pain score on visual analogue scale

Study or subgroup	Spinal N	Mean(SD)	General N	Mean(SD)	Di <sup>.</sup> IV,Fi»	Mean fference ked,95% C	1	Weight	Mean Difference IV,Fixed,95% Cl
Lertakyamanee 1999	118	0.76 (2.04)	103	0.07 (0.34)		+		100.0 %	0.69 [ 0.32, 1.06 ]
Total (95% CI)	118		103			•		100.0 %	0.69 [ 0.32, 1.06 ]
Heterogeneity: not applica	ble								
Test for overall effect: Z =	3.62 (P = 0	.00030)							
Test for subgroup difference	es: Not app	olicable							
					-10 -5	0 5	10		
					Favours spinal	Favou	ırs general		

# Analysis 3.1. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 1 Umbilical arterial pH.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 3 Combined spinal-epidural versus general anaesthesia

Outcome: I Umbilical arterial pH

Study or subgroup	Combined regional N	Mean(SD)	General anaesthesia N	Mean(SD)	IV,	Diffe Fixe	Mean erence ed,95% CI	Weight	Mean Difference IV,Fixed,95% CI
Petropoulos 2003	78	7.26 (0.06)	80	7.29 (0.02)		•		78.7 %	-0.03 [ -0.04, -0.02 ]
Wallace 1995	27	7.27 (0.05)	26	7.3 (0.05)		-	ł	21.3 %	-0.03 [ -0.06, 0.00 ]
<b>Total (95% CI)</b> Heterogeneity: Chi <sup>2</sup> = Test for overall effect: Test for subgroup diffe	<b>105</b> = 0.00, df = 1 (P = 1.0 Z = 4.73 (P < 0.000) erences: Not applicab	00); I <sup>2</sup> =0.0% 01) Ie	106			•		100.0 %	-0.03 [ -0.04, -0.02 ]
				-C Fav	0.5 -0.25 vours genera	( 1	0 0.25 ( Favours con	0.5 nbined	

### Analysis 3.2. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 2 Apgar score of 6 or less at 1 minute (not prespecified in protocol).

Review: Regional versus general anaesthesia for caesarean section

Comparison: 3 Combined spinal-epidural versus general anaesthesia

-

-

Outcome: 2 Apgar score of 6 or less at 1 minute (not prespecified in protocol)

Study or subgroup	Combined regional	General anaesthesia			Risk Ratio		Weight	Risk Ratio
	n/N	n/N		M-H,Fi	ixed,95% Cl			M-H,Fixed,95% CI
Petropoulos 2003	4/78	5/80		-	-		49.2 %	0.82 [ 0.23, 2.94 ]
Wallace 1995	5/27	5/26		_	<b>-</b>		50.8 %	0.96 [ 0.32, 2.94 ]
Total (95% CI)	105	106		-	•		100.0 %	0.89 [ 0.38, 2.07 ]
Total events: 9 (Combine	ed regional), 10 (General ar	naesthesia)						
Heterogeneity: $Chi^2 = 0$	0.03, df = 1 (P = 0.85); $I^2 = 0$	).0%						
Test for overall effect: Z	= 0.26 (P = 0.79)							
Test for subgroup differe	nces: Not applicable							
			0.01	0.1	1 10	100		

Favours combined Favours general

### Analysis 3.3. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 3 Apgar score of 6 or less at 5 minutes (not prespecified in protocol).

Review: Regional versus general anaesthesia for caesarean section

Comparison: 3 Combined spinal-epidural versus general anaesthesia

Outcome: 3 Apgar score of 6 or less at 5 minutes (not prespecified in protocol)

Study or subgroup	Combined regional	General anaesthesia	F	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fix	ed,95% Cl		M-H,Fixed,95% Cl
Petropoulos 2003	2/78	3/80			59.2 %	0.68 [ 0.12, 3.98 ]
Wallace 1995	1/27	2/26	<b>←</b>		40.8 %	0.48 [ 0.05, 4.99 ]
Total (95% CI)	105	106			100.0 %	0.60 [ 0.15, 2.44 ]
Total events: 3 (Combine	ed regional), 5 (General ana	esthesia)				
Heterogeneity: $Chi^2 = 0$	.06, df = 1 (P = 0.81); $I^2 = 0$	).0%				
Test for overall effect: Z	= 0.71 (P = 0.48)					
Test for subgroup differe	nces: Not applicable					
			0.1 0.2 0.5	2 5 10		
			Favours combined	Favours general		

Regional versus general anaesthesia for caesarean section (Review)

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

# Analysis 3.4. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 4 Need for oxygen therapy or mask ventilation of neonate.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 3 Combined spinal-epidural versus general anaesthesia

Outcome: 4 Need for oxygen therapy or mask ventilation of neonate

-

-

Study or subgroup	Combined regional	General anaesthesia	F	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fi>	ked,95% Cl		M-H,Fixed,95% CI
Petropoulos 2003	10/78	9/80			100.0 %	1.14 [ 0.49, 2.65 ]
Total (95% CI)	78	80			100.0 %	1.14 [ 0.49, 2.65 ]
Total events: 10 (Combin	ned regional), 9 (General an	aesthesia)				
Heterogeneity: not appli	cable					
Test for overall effect: Z	= 0.30 (P = 0.76)					
Test for subgroup differe	nces: Not applicable					
			0.1 0.2 0.5	2 5 10		
			Favours combined	Favours general		

# Analysis 3.5. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 5 Mean Apgar score at 1 minute.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 3 Combined spinal-epidural versus general anaesthesia

Outcome: 5 Mean Apgar score at 1 minute

Study or subgroup	Combined regional		General anaesthesia		Di	Mean fference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Fi:	ked,95% Cl		IV,Fixed,95% CI
Korkmaz 2004	15	8.25 (0.62)	15	8 (0.45)		-	100.0 %	0.25 [ -0.14, 0.64 ]
Total (95% CI)	15		15			•	100.0 %	0.25 [ -0.14, 0.64 ]
Heterogeneity: not a	pplicable							
Test for overall effect	: Z = 1.26 (P = 0.21)							
Test for subgroup dif	ferences: Not applicab	le						
				1		<u> </u>		
				-1	) -5	0 5	10	
				Fav	ours general	Favours o	ombined	

## Analysis 3.6. Comparison 3 Combined spinal-epidural versus general anaesthesia, Outcome 6 Mean Apgar score at 5 minutes.

Review: Regional versus general anaesthesia for caesarean section

Comparison: 3 Combined spinal-epidural versus general anaesthesia

Outcome: 6 Mean Apgar score at 5 minutes

Study or subgroup	Combined regional		General anaesthesia			Diffe	Mean erence		Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)		IV,Fixe	ed,95% C	I	IV,Fixed,95% CI
Korkmaz 2004	15	10 (0)	15	9.91 (0.3)					0.0 [ 0.0, 0.0 ]
Total (95% CI)	15		15						0.0 [ 0.0, 0.0 ]
Heterogeneity: not app	olicable								
Test for overall effect: 2	Z = 0.0 (P < 0.0001)								
Test for subgroup diffe	rences: Not applicable								
							· · ·		
					-10 -	5	0 5	10	
					Favours ger	neral	Favou	rs combine	d

### ADDITIONAL TABLES

Table 1. Umbilical artery pH

Study ID	(Spinal) Number	Median	Range	(GA) Number	Median	Range
Dyer 2003	34	7.2	6.93-7.34	32	7.23	7.05-7.40

### APPENDICES

### Appendix I. Search strategy

For the first version of the review, authors searched CENTRAL (*The Cochrane Library* 2005, Issue 1) using the terms general, regional, spinal, epidural, caesarean section, cesarean section and also MEDLINE (1966 to December, 2005) and EMBASE (1980 to December, 2005) using the following strategy:

- 1. general
- 2. regional
- 3. spinal
- 4. epidural
- 5. #1 and (#2 or #3 or #4)
- 6. anaesthesia
- 7. anaesthesia
- 8. #5 and (#6 or #7)
- 9. caesarean section
- 10. cesarean section
- 11. #8 and (#9 or #10)
- 12. random\*
- 13. controlled-clinical-trial
- 14. #12 or #13
- 15. #11 and #14

#### Appendix 2. Methods used to assess trials included in previous versions of this review

Bosede Afolabi (BA) selected potentially relevant trials from those identified by the search strategy and retrieved the full articles. She ensured that multiple publications from the same data set were only used once. BA and Afolabi Lesi (AL) independently assessed each trial for inclusion in the review using the information described in the section 'Criteria for considering studies for this review'. Studies that did not meet the inclusion criteria were excluded and the reason was stated in the table of 'Characteristics of excluded studies'.

BA and AL independently assessed the methodological quality of the included trials. Generation of allocation sequence, allocation concealment, blinding and loss to follow-up are the quality components that were used. For each trial, each quality component apart from blinding was classed as adequate, inadequate or unclear (Juni 2001). For allocation concealment, the letters A to D were used: where A = adequate, B = unclear, C = inadequate and D = not used. For loss to follow-up, inclusion of 90% of participants was considered adequate. Blinding was assessed using the following criteria: blinding of participants, blinding of caregiver and blinding of outcome assessment. Blinding was assessed as open or single blind. Disagreements were resolved by discussion. Where the method used was unclear, the trialists were contacted to clarify the issue. Nkihu Merah (NM) helped resolve disagreements, commented on and helped revise the draft of the review.

BA and AL extracted data from each included trial independently. BA entered data into Review Manager (RevMan 2003). For binary outcomes we recorded the number of participants experiencing the event in each group of the trial. For continuous outcomes for

each group we extracted information to allow calculation of arithmetic means and standard deviations. If the data were reported using geometric means, we extracted information to calculate standard deviations on the log scale. Medians and ranges were extracted and reported in tables. Statistical analyses were carried out using the Review Manager software (RevMan 2003). Binary data were presented as odds ratio. For continuous data, we used the mean difference.

We assessed heterogeneity amongst trials by inspecting the forest plots and using the I-squared test for heterogeneity, where a figure greater than 50% indicates substantial heterogeneity.

We explored the following potential source(s) of heterogeneity using subgroup analysis:

- 1. elective and emergency caesarean section;
- 2. different criteria for the use of the terms 'elective' and 'emergency' caesarean section;
- 3. different indications for caesarean section.

After including all eligible studies in the primary analysis, we conducted sensitivity analyses for each of the quality factors, where possible, using the subgroups adequate, inadequate, or unclear. We also conducted sensitivity analyses for the different outcome criteria.

## WHAT'S NEW

Last assessed as up-to-date: 30 November 2011.

Date	Event	Description
30 November 2011	New search has been performed	Search updated. Eight new studies identified: Six have been included (Jain 2009; Kim 2000; Mancuso 2010; Moslemi 2007; Nabhan 2009; Yentur 2009); one has been excluded (Fyneface-Ogan 2008) and one is await- ing classification (Waris 2002). From the studies previously awaiting classification: Eight studies have now been included (Akyol 2006; Braithwaite 1993; Dermitzaki 2009; Dogan 2008; Mathur 2002; Momani 2001; Papadopoulou 2005; Turhanoglu 1999) and one study has now been ex- cluded (Kamat 1991). Two studies that were previously included as sepa- rate studies (Lertakyamanee 1999 and Kolatat 1999) have now been combined under one study identifier (Lertakyamanee 1999) because they are part of the same study. This updated review is now comprised of 29 included studies, nine excluded studies We updated the search on 20 August 2012 and added a further six reports to the one already in Studies await- ing classification for consideration in the next update The methods have been updated.
1 July 2011	New citation required but conclusions have not changed	This review has been updated. The conclusions have not changed

## HISTORY

Protocol first published: Issue 3, 2003

Review first published: Issue 4, 2006

Date	Event	Description
1 October 2009	Amended	Search updated. Eight reports added to Studies awaiting classification (Akyol 2006a; Dermitzaki 2009a; Dogan 2008a; Kamat 1991a; Mathur 2002a; Momani 2001a; Papadopoulou 2005a; Turhanoglu 1999a)
25 June 2008	Amended	Corrected minor data entry reported by Charles Agert. The event counts for Wallace 1995 had been swapped for comparison 3.02 and 3.03.
8 May 2008	Amended	Converted to new review format.

### CONTRIBUTIONS OF AUTHORS

For the 2011 update, Bosede Afolabi and Afolabi Lesi both extracted the data from the relevant trials and entered the data into RevMan 2011. Bosede Afolabi and Afolabi Lesi updated the Results and Bosede Afolabi updated the Background and Discussion sections.

Bosede Afolabi developed and wrote the protocol. She also extracted data from relevant trials, entered the data into Review Manager, and co-wrote the first version of this review. Afolabi Lesi commented on and revised the draft of the protocol during its development. He also extracted data and co-wrote and revised the review.

## DECLARATIONS OF INTEREST

None known.

## DIFFERENCES BETWEEN PROTOCOL AND REVIEW

The methods have been update to reflect the latest *Cochrane Handbook for Systematic Reviews of Interventions* (Higgins 2011). Our outcomes have been separated into 'primary' and 'secondary' outcomes.

We have also added a number of outcomes which were not prespecified in the protocol:

### **Primary outcomes**

#### Maternal

- Amount of blood transfusion received in units (not prespecified in protocol)
- Number who received postoperative blood transfusion (not prespecified in protocol)

### Secondary outcomes

### Maternal

• Time to request postoperative analgesia in minutes (not prespecified in protocol)

• Adverse events such as anaphylactic reactions, thromboembolic disease and backache. Headache, epigastric pain, blurred vision, convulsions, pruritus, shivering and bradycardia were also measured despite not being prespecified in the protocol

#### Neonatal

- Apgar score of four or less at one and five minutes (not prespecified in protocol)
- Apgar score of six or less at one and five minutes (not prespecified in protocol)
- Apgar score of eight or less at one and five or 10 minutes (not prespecified in protocol
- Mean neonatal Apgar scores at one and 10 minutes (not prespecified in protocol)

## INDEX TERMS

### Medical Subject Headings (MeSH)

\*Anesthesia, Conduction; \*Anesthesia, General; \*Cesarean Section; Anesthesia, Obstetrical [\*methods]; Blood Loss, Surgical [statistics & numerical data]; Hematocrit; Randomized Controlled Trials as Topic

### MeSH check words

Female; Humans; Pregnancy